

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Altindag, Onur; Erten, Bilge; Keskin, Pinar

Working Paper Mental Health Costs of Lockdowns: Evidence from Age-Specific Curfews in Turkey

IZA Discussion Papers, No. 14281

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Altindag, Onur; Erten, Bilge; Keskin, Pinar (2021) : Mental Health Costs of Lockdowns: Evidence from Age-Specific Curfews in Turkey, IZA Discussion Papers, No. 14281, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/236312

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 14281

Mental Health Costs of Lockdowns: Evidence from Age-Specific Curfews in Turkey

Onur Altindag Bilge Erten Pinar Keskin

APRIL 2021



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 14281

Mental Health Costs of Lockdowns: Evidence from Age-Specific Curfews in Turkey

Onur Altindag

Bentley University

Bilge Erten Northeastern University and IZA

Pinar Keskin Wellesley College

APRIL 2021

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Mental Health Costs of Lockdowns: Evidence from Age-Specific Curfews in Turkey^{*}

Using a strict, age-specific lockdown order for adults aged 65 and older in Turkey, we examine the mental health consequences of an extended period of tight mobility restrictions on senior adults. Adopting a regression discontinuity design, we find that the curfew-induced decline in mobility substantially worsened mental health outcomes, including somatic and nonsomatic symptoms of mental distress (approximately 0.2 standard deviation). Exploring potential channels, we document an increase in social and physical isolation, with no evidence of robust changes in labor market outcomes or intrahousehold conflict for this subpopulation.

JEL Classification:	I18, I31, O15
Keywords:	COVID-19, mental health, lockdown, regression discontinuity,
	Turkey

Corresponding author: Bilge Erten

Department of Economics 43 Leon Street 312A Lake Hall Northeastern University Boston, MA 02115 USA E-mail: b.erten@northeastern.edu

^{*} We gratefully acknowledge financial support from the Northeastern University CSSH COVID-19 Research and Engagement Development Initiative. We thank Emilie Courtin, James Dana, Donghee Jo, Phillip Levine, Alisa Lincoln, Mindy Marks, Stephen O'Connell, Olga Shurchkov, Berna Turam, Jeff Gulati, and the webinar participants at Ankara University, Bentley University, Grand Valley State University, LACDEV, the NBER Health Economics Meeting Fall 2020, NEUDC, Northeastern University, Washington and Lee University, and Wellesley College for their useful comments and suggestions. This study was approved by the Northeastern University IRB (20-05-12). All errors are, of course, our own.

Mental health conditions account for 20 percent of all disabilities worldwide and cost approximately more than 1 trillion USD annually (World Health Organization 2019). Social isolation and loneliness have shown strong correlations with depression and anxiety and can significantly predict adult morbidity and chronic diseases that lead to early mortality (Idler and Benyamini 1997; Ortega et al. 2010; Steptoe et al. 2013; Holwerda et al. 2016; Domènech-Abella et al. 2017).

Recent anecdotal evidence suggests a considerable increase in mental health disorders after the introduction of movement restrictions to slow the spread of COVID-19 (The New York Times May 12, 2020). The share of the U.S. population who report symptoms of depression and anxiety, for example, rose to around 40 percent during the COVID-19 pandemic compared to 11 percent in early 2019 (Center for Disease Control and Prevention 2020).¹ Stay-at-home orders are associated with an increased number of internet search queries related to mental health, most strikingly suicide intentions (Jacobson et al. 2020).² Mental health effects are expected to worsen over time as self- or government-imposed quarantine and other social distancing measures create increased social isolation, physical immobility, and economic uncertainty. The financial cost of treating COVID-19 related mental health conditions is staggering: one recent estimate, for example, suggests a 1.6 trillion USD additional annual burden to the U.S. healthcare system (Cutler and Summers 2020). Despite the mounting descriptive evidence coupled with increasing concern among public health and medical communities (Armitage and Nellums 2020; Brooks et al. 2020; Galea et al. 2020; Holmes et al. 2020; Panchal et al. 2020), scant empirical evidence exists to show the causal impact of restricted mobility on mental health outcomes.

We help fill this gap by quantifying the effects of binding stay-at-home orders for those aged 65 and older implemented in Turkey on individuals' mental health. The Turkish government imposed a strict stay-at-home order for the high-risk population of those 65 and older starting on March 21, 2020. Severe financial penalties were imposed for noncompliance with these mandated movement restrictions. The strict curfew orders remained in effect until June 2020, making it one of the longest confinement policies enacted worldwide to reduce COVID-19 mortality.

¹In Spain, similarly, based on a cross-sectional survey, González-Sanguino et al. (2020) report psychological stress among 87 percent of survey participants and attribute its excessive prevalence to confinement.

²Similarly, individuals exposed to stay-at-home orders in the United States report increased health concerns, financial anxiety, and loneliness compared to those who are not exposed to them (Tull et al. 2020).

Our study exploits the binding age cutoff to causally estimate the impact of the agespecific curfew on mental health outcomes. We rely on the fact that individuals who were born around the age cutoff at which the curfew becomes binding have no systematic differences in predetermined characteristics and are thus comparable. To this end, we conducted a phone survey from late May to early July, targeting the specific age group of 59- to 70-year-old adults, and compared those who were just below the age cutoff of 65 and thus not affected by the stay-at-home order to those just above 65 who were affected. In addition to using the 20-item Self Reporting Questionnaire (SRQ-20) developed by the World Health Organization (WHO 1994)—a validated mental health measure used in a variety of nonclinical settings—we designed a survey to capture various channels through which the curfew can impact mental health outcomes.

Social isolation may generate adverse mental health consequences through several different channels we examined in this study. First, stay-at-home orders reduce one's contact with other individuals and reduce social interactions, leading to increased feelings of loneliness and anxiety. Since older adults are already at risk for depression and mental illness, being prohibited from seeing their close relatives or friends is likely to act as an additional stressor, making them feel lonelier, more anxious, and forgotten (Armitage and Nellums 2020; Santini et al. 2020; Newman and Zainal 2020). Second, stay-at-home orders may prevent individuals from participating in the workforce and reduce their potential income, leading to additional stress due to financial constraints (Fetzer et al. 2020a; Beland et al. 2020).^{3,4} Third, being confined at home with other family members for an extended time period can also increase the likelihood of intrahousehold conflict, and in extreme cases, can give rise to physical or psychological abuse (Leslie and Wilson 2020; Ravindran and Shah 2020).⁵

⁵Although the incidence of domestic violence appears smaller among older adults, it is far from negligible, ranging from around 3 to 10 percent (Nelson 2002; Tufan 2011).

³In the U.S. context, negative effects on labor market outcomes have occurred (Forsythe et al. 2020). These studies also document a smaller and imprecisely estimated effect for the labor market outcomes of older adults (Gupta et al. 2020). Similarly, since we focus on a relatively older segment of the population, a large proportion of which is already out of the labor force, one might expect to see smaller effects on labor market outcomes.

⁴The retirement age in Turkey is 58 for women and 60 for men. For new entrants to the pension system after October 2008, the retirement age will gradually rise to 65 (OECD 2017). Hence, our respondents were not differentially affected by the retirement age cutoff. Using the HLFS 2019, Figure A1 illustrates graphically that the probability of retirement increases monotonically from age 63 to 68. Note that the HLFS 2019 does not contain birth month information, which is necessary to conduct an RD analysis. It is also important to remember that no specific government programs in Turkey exist that are similar to Medicare in the US for which individuals qualify once they turn 65.

We adopt a regression discontinuity design (RD) using a narrow age bandwidth and report three main results. First, our RD estimates show that the curfew reduced the number of days that individuals had gone outside in the previous week by around one day, corresponding to an approximately 47 percent decline relative to the control group. Similarly, it increased the probability of never leaving home by approximately 30 percentage points, corresponding to a 149 percent increase relative to the control group.

Second, we find that the curfew-induced reduction in mobility increases the probability of experiencing mental distress markedly, measured both by somatic indicators that capture physical symptoms of anxiety and depression and nonsomatic indicators that represent more subjective assessments of anxiety and depression. Our RD estimates imply that exposure to the curfew results in a 0.18 standard deviation increase in somatic symptoms and a 0.16 standard deviation increase in nonsomatic symptoms of mental distress.⁶

Lastly, examining potential channels, we document that social and physical isolation play a particularly important role in explaining our results. Our results indicate that exposure to the curfew results in a 8.8 percentage point (16 percent) increase in the probability of having limited social interaction with friends and family, and a 24.8 percentage point (54 percent) increase in the probability of having limited physical activity. We find no evidence of a robust significant change in labor market outcomes or intrahousehold conflict measures.

We note upfront that the order of questions asked in the survey may lead to potential priming effects. In particular, the respondents were asked about their mobility indicators prior to their mental health outcomes. If this order of questions makes mobility restrictions more salient, such priming could lead to a change in the respondents' reported mental health measures. However, it is important to note that there were a range of other questions in between mobility and mental health questions as shown in Appendix C, which potentially mitigate such concerns. Adding these transition questions might offset any priming effects.⁷

We make several contributions to the existing literature. First, we show that the adverse

⁶These effect sizes are similar to those reported in studies that document the substantial effects of cash transfers on psychological well-being and depression (Baird et al. 2013; Haushofer and Shapiro 2016).

⁷In surveys to measure Gallup-Sharecare Well-Being Index, it was observed that when subjective wellbeing questions are asked after political preference questions, the respondents reported substantially lower life satisfaction. In a randomized setting, Deaton (2011) shows that adding a single transition question between the political and subjective well-being questions almost fully mitigates the priming effects of political questions. The study suggests that the relevant mechanism is reorientating the respondent's attention away from politics.

impacts of social and physical isolation on mental health are substantial. Despite the wellknown associations, most of the previous studies that document adverse mental health effects stemming from quarantine and social isolation are based on small sample sizes and fail to account for reverse causality (Brooks et al. 2020; Newman and Zainal 2020). As unobservables, such as earlier life events, childhood circumstances, and ability, might affect both social isolation and mental health outcomes, establishing a causal relationship has been difficult. Our empirical setup allows us to estimate the effects of an exogenous decline in mobility on somatic and nonsomatic mental distress indicators.

Our study more closely relates to a smaller subset of empirical studies that rely on the differential timing of lockdown measures.⁸ Using two waves of cross-section data from the U.S., Adams-Prassl et al. (2020) compares the respondents from states that imposed strict stay-at-home orders to those that did not before and after the lockdown. Both Brodeur et al. (2020) and Tubadji et al. (2020) use Google Trends data from the US and Europe and compare the intensity of searching for mental health terms before and after a lockdown. While these studies suggest increased mental distress related to lockdowns, they face challenges in separately identifying the impact of mandated lockdowns from the private responses to COVID-19 cases. Our study provides an empirical setup that isolates the effects of mandated curfews and uses validated measures of mental health.

Second, our findings contribute to better understanding the costs associated with lockdowns, which go beyond financial losses. The growing literature on the optimal policy response to the pandemic often uses a susceptible-infectious-recovered (SIR) framework, assuming that different subpopulations might have different rates of infection and survival (Acemoglu et al. 2020; Alvarez et al. 2020; Brotherhood et al. 2020). Acemoglu et al. (2020), for instance, suggest that it is possible to achieve better outcomes through a simple "targeted policy that applies an aggressive lockdown" on individuals above age 65. If the policy response to COVID-19 creates a mental health crisis by placing already susceptible populations at higher risk of depression and suicide, these consequences would call for additional policy interventions to address such adverse effects. Such policy measures may include setting up mental health call centers, improving access to telehealth services, and

⁸We provide a detailed table in the Appendix A that summarizes some of the most recent studies on this topic (Adams-Prassl et al. 2020; Armbruster and Klotzbücher 2020; Banks and Xu 2020; Burdett et al. 2020; Daly et al. 2020; Etheridge and Spantig 2020; Fetzer et al. 2020a,b; Giuntella et al. 2020; Holman et al. 2020; Proto and Quintana-Domeque 2020).

establishing on-the-ground local support services for at-risk populations (Galea et al. 2020).

This paper is organized as follows. Section 1 provides a brief description of the COVID-19 lockdown in Turkey. Section 2 presents the data used for the analysis, the identification strategy, and preliminary checks for the RD analysis. Section 3 presents the main results, and Section 4 discusses the evidence on potential causal channels. Section 5 concludes the paper.

1 Background

The Turkish Ministry of Health reported the first case of the novel coronavirus on March 10, 2020, and the first COVID-19-related death on March 17.⁹ From this early period, the older population and individuals with underlying medical conditions were defining features of the government's response to the COVID-19 pandemic. In stark contrast to the rest of the world, the Turkish government imposed strict and long-lasting mobility restrictions exclusively on senior citizens. The first curfew decree was issued on March 21 and imposed an absolute lockdown on individuals aged 65 and older and those with certain health conditions.¹⁰ The central government formed local support teams to provide basic needs for individuals subject to the decree and who needed assistance, while no exceptions existed that could breach the stay-at-home order.¹¹ The age-specific curfew along with other government measures to contain the virus, such as mask wearing in public spaces, were routinely enforced by the local security forces and the offenders were fined.^{12, 13}

Only after May 10, individuals who were subject to the curfew were allowed a period of four hours to walk outside their home on Sundays, which was conditional on wearing a mask and social distancing.¹⁴ One week later, the government allowed a similar exception

[%]https://covid19.saglik.gov.tr/ - last accessed April 9, 2021.

¹⁰These conditions include autoimmune disorders, chronic pulmonary disease, asthma, cardiovascular disease, hypertension, renal, and liver-related diseases, https://www.icisleri.gov.tr/65-yas-ve-ustu-ile-kronik-rahatsizligi-olanlara-sokaga-cikma-yasagi-genelgesi-last accessed April 9, 2021.

¹¹In case of a health emergency, those aged 65 and older could call the emergency number 112, and an ambulance could be sent to their premises for transport to the closest hospital or health clinic for emergency treatment.

¹²According to the law, the fine for curfew offenders was set between 789-3,180 Turkish Liras (https://blog.lexpera.com.tr/bulasici-hastaliklara-iliskin-tedbirlere-aykiri-davranma-sucu-tck-m-195/ – last accessed April 9, 2021). Anecdotal evidence suggests that the upper limit was used to deter potential offenders (see, for example, https://www.hurriyet.com.tr/gundem/sokaga-cikma-yasagina-ragmen-kahvehanedeoyun-oynarken-yakalandilar-41492692 - last accessed April 9, 2021.) As a reference, the minimum monthly wage in Turkey during the same period was 2,943 Turkish Liras.

¹³We also note that the lockdown policy did not place any restrictions on household visitors.

¹⁴https://www.goc.gov.tr/65-yas-ve-uzeri20-yas-altikronik-rahatsizligi-bulunan-kisilerin

for six hours.¹⁵ On May 21, senior citizens were allowed to travel to a specific location, which was conditional on staying for at least one month and not leaving their new shelter.¹⁶ On May 29, actively employed senior citizens were exempt from the lockdown. Finally, on June 10, the curfew was relaxed and all individuals who were subject to it were allowed to be outside between 10 a.m. and 8 p.m.

The heterodox policy response to the pandemic has stirred controversy, as a lack of empirical evidence failed to demonstrate that the decision to impose age-specific curfews would slow down the death toll or virus spread.¹⁷ The Turkish Medical Association (TMA), for example, argues that the excessive restrictions on senior citizens' mobility adversely affected their mental health, severely disrupted their daily routines, and created a sense of unfairness among those under lockdown.¹⁸ According to the TMA, the policy lacks the epidemiological evidence to show its effectiveness in protecting vulnerable populations at the expense of their mental well-being.¹⁹

In official announcements, the Turkish Ministry of Internal Affairs does not mention a specific birthday cutoff for the curfew and uses the term "age 65 and older" to indicate the senior population that is subject to lockdown, although anecdotal evidence suggests that birth year is the sole determinant.²⁰ To confirm the threshold in our sample, we directly asked respondents whether they were subject to the age-specific curfew the government imposed. We then ran a simulation in which we split our analysis sample into treatment and control groups using each birth year and month as the curfew threshold to estimate the average difference in exposure to the curfew between the treatment and control groups

⁻sokaga-cikma-kisitlamasi-istisnasi-genelgesi-merkezicerik - last accessed April 9, 2021

¹⁵https://www.icisleri.gov.tr/65-yas-ve-uzeri-ile-kronik-rahatsizligi-olan-vatandaslarin -sokaga-cikma-gun-ve-saatleri -last accessed April 9, 2021

¹⁶https://www.icisleri.gov.tr/81-il-valiligine-65-yas-ve-uzeri-vatandaslarimiz-icin-seya hat-izin-belgesi-genelgesi-last accessed April 9, 2021

¹⁷The Turkish government does not provide detailed and consistent epidemic data, thus to the best of our knowledge, no empirical studies confirm or refute the success of the age-specific curfew policy. In addition, the reported aggregate figures on deaths substantially underestimate the total case and death toll; one study showed that excess mortality is at least twice as high as the official government death counts due to COVID-19 (Altindag 2020).

¹⁸https://www.ttb.org.tr/415yi6z – last accessed April 9, 2021

¹⁹According to the Ministry of Health, the total number of confirmed COVID-19 cases was 583 in 100,000 for 50–64-year-old individuals and 553 in 100,000 for 65–79-year-olds between June 1 and June 18, 2020. The corresponding death rate for all confirmed cases was 3.19 percent for the former group while it was 13.0 percent for the latter one.

²⁰Separate curfews were imposed on individuals aged 18 and 20, and government announcements indicate that year of birth determines inclusion in that age group. See, for example, https://www.icisleri.gov.tr/ sokaga-cikma-yasagi-bulunan-18---20-yas-arasindaki-genclerle-ilgili-istisnalar - last accessed April 9, 2021.

for each of the simulated thresholds.

As shown in Figure A2 and in line with field observations, we obtain the greatest difference in being subject to the curfew between individuals who were born just before and after January 1956. The estimated coefficient indicates that individuals born before or in December 1955 are 85 percentage points more likely to claim to be subject to the curfew than those who were born in January 1956 or later. In the empirical analysis, we rely on this threshold, which provides the strongest discontinuity in exposure to the treatment, as shown by various measures in Figure 2.

2 Data and Empirical Methodology

2.1 Data

We use a unique dataset covering individuals born in Turkey between 1950 and 1961. The data were collected by KONDA Research and Consultancy, a reputable research and consultancy firm in Istanbul, Turkey. Since the firm regularly conducts nationally representative surveys to provide information on public opinion on a wide range of political issues, they have built a surveyor base throughout the country. Their regular surveys—called KONDA Barometer surveys—are conducted 11 times per year and have successfully predicted election outcomes in recent general elections. Given their record and well-respected position in both the Turkish²¹ and international media²² (The Economist 2008, Reuters 2011, The Economist 2019), we contacted KONDA to implement our survey instrument.

Specifically, we approached KONDA to collect survey data from their existing respondent database with the following two criteria: (i) respondents should reside in urban areas where the curfews are strictly imposed, and (ii) they should be aged between 59 and 70 to have 6 treatment and 6 control cohorts on each side of the curfew threshold. Consequently, the survey instrument was implemented in urban areas across 26 regions from May 29 to July 4 through phone interviews.²³ The average response rate for our survey was 88

²¹https://www.hurriyet.com.tr/gundem/hangi-anket-sirketi-secimlerin-sonucunu-dogru-bildi -29224184

²²See, for example, https://www.economist.com/briefing/2008/07/17/flags-veils-and-sharia?sto ry_id=E1_TTSQVVSD, https://www.reuters.com/article/us-turkey-referendum-poll/poll-shows-bac king-for-turk-reforms-on-eve-of-vote-idUSTRE68A0EV20100911?feedType=RSS&feedName=everythi ng&virtualBrandChannel=11563, https://www.economist.com/erasmus/2019/07/01/in-turkey-demogra phy-is-a-brake-on-islamisation.

²³On average, the respondents were subject to the curfew for 8 to 9 weeks when they were contacted.

percent. Although response rates are slightly lower for those not exposed to the curfew, Figure A3 shows no evidence of a significant break in nonresponse rates around the age threshold of 65.²⁴ Appendix C provides the full list of questions asked in the survey.

Appendix Table A1 compares basic demographic information from our analysis sample to the 2019 Household Labor Force Survey (HLFS), focusing on individuals born between 1950 and 1961. We observe that the average age is 64 for both samples and that the marital status indicators are quite similar. Our analysis sample has relatively fewer women. It is also composed of more educated individuals than the HLFS due to the urban sampling frame.

Appendix Table 1 presents the summary statistics for our analysis sample composed of a maximum of 1909 individuals. We observe that 27 percent of the sample completed high school or above. The household size prior to the COVID-19 outbreak was 3.3, with approximately 11 percent of the respondents having ever received psychological support and 57 percent having a chronic disease.²⁵ We observe that 48 percent of the respondents in our sample reported being subject to the curfew and the average number of days spent outside in the previous week was 1.9 days.

Finally, we observe that 13 percent worked for pay, and 15 percent worked either for pay or in family businesses. In addition, 14 percent had a job that they could not attend in the previous week. Approximately 60 percent experienced limited social interaction, and 55 percent experienced limited physical activity.²⁶ Their current household size was 3.4, and approximately 37 percent reported having a conflict with a household member over the last month.

To assess mental health outcomes, a set of mental health screening tools have been developed in nonclinical settings. These range from depression scales such as the Beck Depression Inventory (Beck et al. 1961) to more general psychological distress scales such as the K10 scale (Kessler et al. 2002). In our survey, we use the 20-Item Self Reporting

²⁴Since we did not have a universal database of telephone numbers that matched the owners' age, it was not possible to collect data through random digit dialing. Instead, we rely on the existing respondent database of KONDA to draw a sample of individuals between ages 59 and 70 living in urban areas. Sampling from an existing respondent database may also explain our relatively high response rate for a phone survey.

²⁵Less than 7 percent of the respondents reported that they were living alone at home.

²⁶Limited social interaction is a dummy variable that takes the value of one if the respondent reported that his/her social interaction with friends and family has been extremely limited or very limited in the last month compared to pre-COVID times. Limited physical activity is a dummy variable that takes the value of one if the respondent reported that his/her physical activity (e.g., walking, running, doing sports, etc.) has been extremely limited or very limited in the last month compared to pre-COVID times.

Questionnaire (SRQ-20) developed by the World Health Organization (WHO 1994). Table 2 details the list of SRQ-20 questions along with their summary statistics. Among the several mental health screening tools, the SRQ-20 is one of the few specifically designed for low- and middle-income settings. The questions are designed to identify mental distress that captures typical anxiety and depression symptoms, such as poor concentration or suicidal thoughts, and less-known somatic symptoms such as digestive problems or frequent aches.²⁷ The questions' short format and the dichotomous answers used in the SRQ-20 render it particularly useful in settings with limited resources (van der Westhuizen et al. 2016).²⁸

Following Anderson (2008) and Erten and Keskin (2020), we construct three summary indices: (i) a mental distress index, which is an average of the z-scores of 20 mental health indicators; (ii) a somatic symptoms of distress index, which is an average of 4 indicators related to the body and are therefore more objective measures of anxiety and depression; and (iii) a nonsomatic symptoms of distress index, which is an average of the remaining 16 indicators that represent more subjective assessments of anxiety and depression. We create these indices to have a mean of 0 and a standard deviation of 1, following Anderson (2008); the variables that compose each index are described in Appendix A. Higher index values reflect higher mental distress levels.

In addition, we include a more standard measure of mental distress using SRQ-20 by summing "yes" answers to the questions included in the SRQ-20 inventory. This variable provides an alternative measure of mental distress, with higher values capturing higher distress levels.²⁹

²⁷The American Psychiatric Association also suggests that anxiety and depression symptoms include not only classic psychological signs such as loss of interest but also somatic symptoms such as general aches and pains or trembling (American Psychiatric Association 2013).

²⁸The SRQ-20 has been cross-validated across many countries, including Brazil (Iacoponi and de Jesus Mari 1989), China (Chen et al. 2009), Vietnam (Giang et al. 2006) and India (Patel et al. 2008), and has been shown to be a reliable tool for measuring mental health distress in low- and middle-income contexts.

²⁹In the psychology literature, some studies used certain cutoff points of the SRQ score varying between 0 and 20 to identify the presence of mental disorders. However, as WHO (1994) has described, it is necessary to conduct an empirical validation against a sample of in-depth psychiatric interviews in order to determine the country- and culture-specific cutoff point (Harpham et al. 2003). Since we do not have a study that conducted such a validation in Turkey, we opted for not using any specific cutoff point.

2.2 Identification

As explained in Section 1, COVID-19 lockdowns were strictly imposed on individuals born before or in December 1955 while those born in January 1956 or later were exempt. The context thus offers an ideal setting to implement an RD design to estimate the curfew's impact on a range of outcomes.

Our RD design leverages the quasi-random assignment of curfew around the age cutoff to estimate the reduced-form (RF) effects of the curfew on our outcomes of interest. The causal interpretation of both estimates relies on the identifying assumption that around the vicinity of the curfew age cutoff, the assignment to curfew is as good as random. Our identifying assumption is that these two cohorts born one month apart do not exhibit any systematic differences other than whether they were exposed to the curfew or not. For our RF estimates, we use the following specification:

$$y_i = \alpha + \beta z_i + f(x_i) + \epsilon_i$$

$$\forall x_i \in (c - h, c + h)$$
(1)

where y_i captures the outcome of interest, which is regressed on a treatment indicator z_i that equals one for individuals who were born before January 1956 and zero otherwise. x_i is the forcing variable defined as the number of months that the respondent is older than the index month of the curfew threshold. The function $f(x_i)$ is a continuous local linear function fit separately on each side of the threshold point c. The standard errors are clustered at the month-year of birth to account for the correlation in outcomes across individuals who were born in the same year-month cell (Lee and Card 2008). We additionally control for month of birth fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. For the regression sample restriction, we use the Imbens and Kalyanaraman (2012) procedure to choose the optimal bandwidth h.³⁰ As discussed in Section 3, our results are robust to a range of bandwidths and a quadratic control function.

To address concerns related to multiple hypothesis tests, we report p-values that account for the false discovery rate (FDR) (Benjamini et al. 2006; Anderson 2008). It is necessary to correct for FDR in our study since we examine multiple outcomes within the same do-

³⁰We use a uniform kernel in our estimations. The results are highly similar when we use a triangular kernel.

main, which increases the likelihood that some of our estimates are statistically significant by chance alone. We use the FDR approach presented in Anderson (2008), based on the methodology in Benjamini et al. (2006). This procedure converts p-values into q-values, which control the expected proportion of rejections that are Type I errors, or false discoveries. Following Field et al. (2021), we make these adjustments for multiple hypothesis testing within domains defined by the set of outcomes considered in each table (e.g., we adjust within all mobility outcomes in Table 3), or each panel within tables (e.g. we adjust within all labor market outcomes in Panel A of Table 5).³¹

One could be concerned that exposure to the curfew affects the perceived likelihood of getting sick in addition to its effects on mobility. However, this concern is highly unlikely given that those just below the cutoff point have heard the same information about COVID-19's higher risks for older people and people with chronic diseases. Nevertheless, if this concern was valid, the estimates capture the combined effect of the lockdown and fear related to COVID-19.

The optimal bandwidth according to Imbens and Kalyanaraman (2012) (IK)'s method is 45 for the variable capturing days spent outside during the previous week. For brevity, we will use the RD estimates for this optimal bandwidth in interpreting the magnitudes of the results. For robustness, we also present our results using the corresponding optimal bandwidth calculated by Calonico et al. (2014) algorithm, which is 17 months. In addition, we use two additional constant bandwidths of 30 and 60 months around the discontinuity in our main tables. The results are highly similar with different bandwidths and control functions.

2.3 Preliminary Checks

We conduct two standard checks to validate our RD design (Imbens and Lemiuex 2008). The first is to test whether the forcing variable is subject to manipulation around the predetermined threshold (McCrary 2008). In our specific setup, rejecting a one-sided null hypothesis would indicate that individuals falsify their birthday to exempt themselves from the curfew. This, however, is highly unlikely because we asked individuals to read their birthday from their national IDs, which is the standard tool that local security forces use

³¹Other standard routines for q-value conversion developed by Simes (1986) and Romano and Wolf (2016) yield similar results, which are available from the authors upon request.

to enforce the curfew. Figure A4 provides visual evidence that the local quadratic fits on the left- and righthand side of the age threshold for the curfew do not exhibit any jumps in observation frequency. A formal test provided in McCrary (2008) also fails to reject the null hypothesis.³²

Second, we examine whether the predetermined covariates are balanced around the discontinuity. In Figure 1, each graph plots local averages of the outcome in one-month bins against the forcing variable. We find no evidence of a significant break at the discontinuity for indicator variables of whether the respondents completed high school, whether they are illiterate, whether they are female, whether they are married, whether they are widowed or separated, whether they are of non-Turkish ethnic origin, whether they have ever received psychological support, whether they have a chronic disease, and their household size prior to COVID-19.³³

3 Effects of the Curfew on Mobility and Mental Health Outcomes

3.1 Mobility Outcomes

We begin by testing the curfew's effect on mobility outcomes. Panel A of Figure 2 plots local averages of three mobility outcomes in monthly bins against the respondent's month and year of birth with a cutoff of December 1955.³⁴ As described in 1, the curfew required those born before this date to stay at home, whereas younger cohorts were free to leave their homes at any time. Local linear smoothed fits on each side of the cutoff are overlaid on each figure. Figure 2(a) shows a clear downward shift at the discontinuity with an approximately 1-day decline in the number of days that respondents went out during the week prior to the interview. Similarly, Figure 2(b) also reveals a clear jump around the discontinuity in the self-reported probability of being subject to the curfew. Similarly, in Figure 2(c), the probability of never going outside—the likelihood of never leaving home—increases abruptly around the age threshold. Compared to the control group averages, all

 $^{^{32}}$ To conduct the test, we use our first-stage optimal bandwidth of 45 months. The test yields a p-value of 0.8.

³³In regression-based tests reported in Appendix Table A2, we note that none of the predetermined covariates display any evidence of a statistically significant jump at the discontinuity across different bandwidths. SUR tests of the coefficients' joint significance result in p-values ranging from 0.25 to 0.75, depending on the bandwidth.

³⁴Appendix Figure A5 provides a residualized version of Figure 2 after regressing outcomes on fixed effects and controls, displaying very similar results.

of these indicators show a substantial decline in the mobility of respondents older than 65 years of age at the time the curfew was imposed.

In Table 3, we present the corresponding first-stage estimates using the main estimating equation at various bandwidths. Crossing the treatment threshold reduces the number of days the respondents went outside in the previous week by 1–1.1 days. The estimates are robust to different bandwidths and roughly correspond to a 47 percent decline in mobility relative to the control group mean.³⁵

Related indicators exhibit similarly large declines in mobility: being born before 1955 increases reporting of being subject to the curfew by 71 percentage points and raises the probability of never going out by 30 percentage points. Relative to the control group means, these estimates correspond to an approximately 7-fold increase in the probability of reporting being subject to the curfew and a 149 percent increase in the probability of never leaving home. These estimates are robust to different bandwidths and control functions, as shown in Appendix Tables A3 and A8.

3.2 Mental Health Outcomes

We next examine the curfew's impact on mental health outcomes. Following recent RD studies³⁶, we graphically illustrate the average of each index as a function of monthly distance from being born in December 1955. In particular, we residualize the outcomes by regressing them on fixed effects and controls, and then plot local averages of these residualized outcomes in monthly bins against the respondent's month and year of birth with a cutoff of December 1955 in Figure 3. These graphs suggest a sharp increase in all mental distress indicators around the discontinuity.³⁷

Table 4 quantifies the magnitude of these effects by reporting the reduced-form RD treatment effects of being born before December 1955 with a linear control function in the month-year of birth on each side of the discontinuity across different bandwidths. Remarkably, the RD estimates show a substantial impact of the curfew on all measures of mental distress; the first-row estimates imply a 0.21 standard deviation increase in

³⁵The results are very similar if we examine the curfew's effects on the number of times the respondent went outside over the last month in a week on average, or the number of times the respondent went outside in the last week apart from the legally permitted times.

³⁶See, for example, Asher and Novosad (2020).

³⁷Appendix Figure A6, which provides an unresidualized version of Figure 3, shows very similar results.

the mental distress index.³⁸ We estimate similar effects for the more objective measure of depression—the somatic symptoms index (0.18 standard deviation)—which includes only physical symptoms of depression. The corresponding effect size of the nonsomatic symptoms index is an increase of 0.16 standard deviation. Finally, the RD estimates indicate that the curfew had a positive impact of 0.7 on the sum of "yes" answers in SRQ-20 inventory reported by the respondents. This corresponds to a 10.4 percent increase relative to the control group mean. These estimates are robust to different bandwidths and control functions, as shown in Appendix Tables A4 and A8.^{39,40}

4 Effects of the Curfew on Potential Causal Channels

We proceed by examining potential causal channels through which the curfew negatively impacted mental health outcomes. We divide our analysis into three subsections by focusing on the curfew's effects on the following outcomes: (i) employment and income, (ii) social and physical isolation, and (iii) household conflict.

4.1 Employment and Income Outcomes

If exposure to the curfew prevents one from going to work, it can negatively impact one's ability to work outside the home and earn a living. Such negative labor market impacts can lead to additional anxiety and a deterioration of mental health outcomes. We explore

³⁸One could consider scaling the magnitudes of these reduced-form estimates by different mobility measures to get a sense of the treatment effects on the complier population. Scaling by reporting that the respondent was under curfew results in larger two-stage-least-squares estimates than the reduced-form estimates since the estimated coefficient for differential likelihood of being under curfew range 0.7 for the optimal bandwidth of 45 months. In particular, complying with the curfew leads to a 0.29 standard deviation increase in the mental distress index. In contrast, scaling with respect to days spent outside last week results in slightly smaller estimates as the reduced-form coefficients are divided by the first-stage estimates of around 1.1 days. For instance, a one-day reduction in days spent outside due to the curfew results in a 0.19 standard deviation increase in mental distress index.

³⁹When we examine the effects of the curfew on individuals' self-perception of their health, Panel A of Appendix Table A5 shows that those exposed to the curfew report a higher probability of having poor physical health compared to those not exposed. Interestingly, we find no evidence that exposure to the curfew has a significant impact on the probability of reporting poor mental health. Hence, despite a significant worsening in their mental distress outcomes as measured by SRQ-questions, those exposed to the curfew do not recognize these symptoms as a deterioration in their mental health.

⁴⁰Moreover, we asked respondents two questions about their life satisfaction using the Cantril Scale, which is frequently used in Gallup Surveys. In reporting scale scores, Gallup refers to those respondents who have poor ratings of their current life situation (4 and below) and negative ratings for the next five years (4 and below) as "suffering", or having well-being that is at high risk (Gallup 2013). In Panel B of Appendix Table A5, we also examine whether the curfew had a significant impact on the probability of reporting higher suffering measured by these indicators, but we find no evidence of a significant change in this measure.

this mechanism by testing whether the curfew negatively impacts employment and income outcomes.

In Panel A of Table 5, we find no evidence that the curfew significantly impacts labor market outcomes or household finances after adjusting for multiple hypothesis testing. These include whether the respondent was working in paid, or paid/unpaid employment, or whether the respondent has a job that he/she cannot attend. Similarly, we find no evidence of the curfew significantly impacting having enough money to meet usual needs, or being worried about spending money.⁴¹ Hence, we conclude that the employment and income channel does not seem to explain our results.

4.2 Social and Physical Isolation Outcomes

Confinement may severely limit an individual's social interaction and physical mobility. Social isolation, loneliness, and disconnectedness from the community may lead to mental health problems among the senior population. Moreover, continuous confinement within the same physical space and a lack of physical mobility and exercise could further magnify the risk of a mental breakdown.

In Panel B of Table 5, the RD estimates show that the curfew had a positive impact on having a very limited or extremely limited social interaction and physical activity in the last month compared to pre-COVID times.⁴² In particular, exposure to the curfew results in a 8.8 percentage point increase in the probability of having limited social interaction with friends and family, corresponding to a 16 percent increase relative to the control group. Similarly, exposure to the curfew leads to a 24.8 percentage point increase in the probability of having, or participating in sports, corresponding to a 54 percent increase compared to the control group. Overall, we conclude that the social and physical isolation channel can potentially explain our results.

4.3 Household Conflict Outcomes

While being confined to the home reduces time spent with people outside of the home, it tends to result in an increase in time spent with household members. This additional time

⁴¹In our survey, we also asked respondents about their monthly income more explicitly. However, many respondents answered this question by reporting that they have retirement income. Since retirement income varies widely across individuals, we do not have a consistent measure of household income.

⁴²The results are very similar if we only focus on experiencing an extreme limitation in social interaction and physical activity over the same time period.

could mechanically increase the probability of having conflict with a household member. Moreover, the additional stress of social isolation could also increase the probability of experiencing a conflict at home.

In Panel C of Table 5, the RD estimates indicate no evidence of change in the current household size, implying no significant impact of the curfew on household composition. We also find no evidence of a significant change in the probability of having a conflict with a household member driven by home confinement. However, we note as a caveat that the precision of estimates for the household conflict effect do not allow us to rule out meaningful results. Overall, the household conflict channel does not appear to explain our results. The graphical illustration of these potential channels presented in Figure A7 are consistent the results shown in Table 5.

Individual beliefs and practices Finally, we explore some potential consequences of the curfew for individual beliefs and practices. For example, age-specific curfews might create a sense of social unfairness among individuals subject to them. As shown in Table 6, individuals subject to the curfew are substantially less likely to support the curfew policy. In particular, exposure to the curfew reduced support for the 65 and over age-specific curfew by 14.6 percent compared to the control group. In contrast, we find no evidence of a robust significant change for satisfaction with the government's overall COVID-19 policy response. Lastly, we examine the change in religious practices and religiosity as a coping mechanism under social isolation. The estimates provided in Appendix Table A7 show no evidence of the curfew significantly impacting religious beliefs and practices.

Note that these estimates are robust to using this study's different bandwidths and control functions as shown in Appendix Tables A6 and A8.

5 Conclusion

The policy response to the COVID-19 pandemic has involved lockdown orders with different degrees of strictness. While the lockdown policies generally brought benefits in reducing the spread of the virus, they also posed potential costs on certain subpopulations. While macroeconomic models incorporating the SIR framework often recommend age-specific lockdowns targeting adults age 65 and older, they often neglect the mental health consequences of these movement restrictions.

Using a rather unique setup in Turkey—which imposed a strict curfew for the high-risk population group of those aged 65 and over on March 21, 2020—we implement an RD design comparing those just under the binding age cutoff to those above it using data from a detailed phone survey covering 59- to 70-year-old adults.

Our findings reveal that the curfew had striking mental health consequences. We find that the curfew reduced the number of days spent outside the week prior to the interview by approximately one day. In turn, the curfew-induced reduction in mobility increases the probability of experiencing mental distress substantially, with approximately 0.2 standard deviation increases in somatic and nonsomatic symptoms of mental distress. These sizable effects are all the more concerning since older adults are already more susceptible to a higher risk of depression and suicide.

These mental health consequences of strict lockdown policies call for a rethinking of how additional policy measures—ranging from mental health call centers and telehealth services to on-the-ground local support for senior adults—can be used to alleviate the mental health burden on susceptible populations.

References

- Acemoglu, Daron, Victor Chernozhukov, Iván Werning, and Michael D Whinston, "A multi-risk SIR model with optimally targeted lockdown," Technical Report, National Bureau of Economic Research 2020.
- Adams-Prassl, Abi, Teodora Boneva, Marta Golin, Christopher Rauh et al., "The Impact of the Coronavirus Lockdown on Mental Health: Evidence from the US," Technical Report 2020.
- Ahmed, Md Zahir, Oli Ahmed, Zhou Aibao, Sang Hanbin, Liu Siyu, and Akbaruddin Ahmad, "Epidemic of COVID-19 in China and associated Psychological Problems," *Asian journal of psychiatry*, 2020, *51*, 102092–102092.
- Altindag, Onur, "What do official COVID19 death statistics tell us?," 2020. Accessed at https://www.onuraltindag.info/pt-tr/posts/excess_mortality_eng/ on August 8, 2020.
- **Alvarez, Fernando E, David Argente, and Francesco Lippi**, "A simple planning problem for covid-19 lockdown," Technical Report, National Bureau of Economic Research 2020.
- **American Psychiatric Association**, *Diagnostic and statistical manual of mental disorders (DSM-5*®), American Psychiatric Pub, 2013.
- **Anderson, Michael L**, "Multiple inference and gender differences in the effects of early intervention: A reevaluation of the Abecedarian, Perry Preschool, and Early Training Projects," *Journal of the American statistical Association*, 2008, *103* (484), 1481–1495.
- Armbruster, Stephanie and Valentin Klotzbücher, "Lost in lockdown? COVID-19, social distancing, and mental health in Germany," Technical Report, Diskussionsbeiträge 2020.
- **Armitage, Richard and Laura B Nellums**, "COVID-19 and the consequences of isolating the elderly," *The Lancet Public Health*, 2020, *5* (5), e256.
- Asher, Sam and Paul Novosad, "Rural Roads and Local Economic Development," American Economic Review, 2020, 110 (3), 797–823.
- **Baird, Sarah, Jacobus De Hoop, and Berk Özler**, "Income shocks and adolescent mental health," *Journal of Human Resources*, 2013, *48* (2), 370–403.
- **Banks, James and Xiaowei Xu**, "The mental health effects of the first two months of lockdown and social distancing during the Covid-19 pandemic in the UK," Technical Report, IFS Working Papers 2020.
- Beck, Aaron T, C Ward, M Mendelson, J Mock, and J Erbaugh, "Beck depression inventory (BDI)," Arch Gen Psychiatry, 1961, 4 (6), 561–571.
- **Beland, Louis-Philippe, Abel Brodeur, Derek Mikola, and Taylor Wright**, "The Short-Term Economic Consequences of COVID-19: Occupation Tasks and Mental Health in Canada," 2020.
- Benjamini, Yoav, Abba M Krieger, and Daniel Yekutieli, "Adaptive linear step-up procedures that control the false discovery rate," *Biometrika*, 2006, 93 (3), 491–507.

- **Brodeur, Abel, Andrew Clark, Sarah Fleche, and Nattavudh Powdthavee**, "Covid-19, lockdowns and well-being: Evidence from google trends," 2020.
- Brooks, Samantha K, Rebecca K Webster, Louise E Smith, Lisa Woodland, Simon Wessely, Neil Greenberg, and Gideon James Rubin, "The psychological impact of quarantine and how to reduce it: rapid review of the evidence," *The Lancet*, 2020.
- **Brotherhood, Luiz, Philipp Kircher, Cezar Santos, and Michèle Tertilt**, "An economic model of the Covid-19 epidemic: The importance of testing and age-specific policies," 2020.
- Burdett, Ashley, Ben Etheridge, and Lisa Spantig, "Well-Being and Weather During Lockdown," 2020.
- Calonico, Sebastian, Matias Cattaneo, and Rocio Titiunik, "Robust Nonparametric Confidence Intervals for Regression Discontinuity Designs," *Econometrica*, 2014, 82, 2295–2326.
- Center for Disease Control and Prevention, "Mental health: Household Pulse Survey," 2020. https://www.cdc.gov/nchs/covid19/pulse/mental-health.htm last accessed 2020-10-31.
- **Chen, Shulin, Guoqiu Zhao, Lingjiang Li, Yiqiang Wang, Helen Chiu, and Eric Caine**, "Psychometric properties of the Chinese version of the Self-Reporting Questionnaire 20 (SRQ-20) in community settings," *International Journal of Social Psychiatry*, 2009, 55 (6), 538–547.
- Cutler, David M. and Lawrence H. Summers, "The COVID-19 Pandemic and the 16 Trillion Virus," *JAMA*, 10 2020, 324 (15), 1495–1496.
- **Daly, Michael, Angelina R Sutin, and Eric Robinson**, "Longitudinal changes in mental health and the COVID-19 pandemic: Evidence from the UK Household Longitudinal Study," *Psychological medicine*, 2020, pp. 1–10.
- **Deaton, Angus**, "The financial crisis and the well-being of Americans: 2011 OEP Hicks Lecture *," *Oxford Economic Papers*, 11 2011, *64* (1), 1–26.
- Domènech-Abella, Joan, Elvira Lara, Maria Rubio-Valera, Beatriz Olaya, Maria Victoria Moneta, Laura Alejandra Rico-Uribe, Jose Luis Ayuso-Mateos, Jordi Mundó, and Josep Maria Haro, "Loneliness and depression in the elderly: the role of social network," Social psychiatry and psychiatric epidemiology, 2017, 52 (4), 381–390.
- **Eghigian, Greg**, "The Spanish Flu Pandemic and Mental Health: A Historical Perspective," Technical Report, Psychiatric Times 2020.
- El-Zoghby, Safaa M, Enayat M Soltan, and Hend M Salama, "Impact of the COVID-19 Pandemic on Mental Health and Social Support among Adult Egyptians," *Journal of community health*, 2020, 45 (4), 689–695.
- Erten, Bilge and Pinar Keskin, "Breaking the cycle? education and the intergenerational transmission of violence," *Review of Economics and Statistics*, 2020, 102 (2), 252–268.
- Etheridge, Ben and Lisa Spantig, "The gender gap in mental well-being during the Covid-19 outbreak: evidence from the UK," Technical Report, Institute for Social and Economic Research 2020.

- **Fegert, Jörg M, Benedetto Vitiello, Paul L Plener, and Vera Clemens**, "Challenges and burden of the Coronavirus 2019 (COVID-19) pandemic for child and adolescent mental health: a narrative review to highlight clinical and research needs in the acute phase and the long return to normality," *Child and adolescent psychiatry and mental health*, 2020, 14, 1–11.
- Fetzer, Thiemo, Lukas Hensel, Johannes Hermle, and Christopher Roth, "Coronavirus perceptions and economic anxiety," *Review of Economics and Statistics*, 2020, pp. 1–36.
- Fetzer, Thiemo R, Marc Witte, Lukas Hensel, Jon Jachimowicz, Johannes Haushofer, Andriy Ivchenko, Stefano Caria, Elena Reutskaja, Christopher P Roth, Stefano Fiorin et al., "Global Behaviors and Perceptions at the Onset of the COVID-19 Pandemic," Technical Report, National Bureau of Economic Research 2020.
- Field, Erica, Rohini Pande, Natalia Rigol, Simone Schaner, and Charity Moore, "On Her Own Account: How Strengthening Women's Financial Control Impacts Labor Supply and Gender Norms," conditionally accepted at American Economic Review, 2021.
- **Forsythe, Eliza, Lisa B Kahn, Fabian Lange, and David Wiczer**, "Labor Demand in the time of COVID-19: Evidence from vacancy postings and UI claims," *Journal of Public Economics*, 2020, pp. 104–238.
- Galea, Sandro, Jennifer Ahern, Heidi Resnick, Dean Kilpatrick, Michael Bucuvalas, Joel Gold, and David Vlahov, "Psychological sequelae of the September 11 terrorist attacks in New York City," *New England journal of medicine*, 2002, 346 (13), 982–987.
- ____, Raina M Merchant, and Nicole Lurie, "The mental health consequences of COVID-19 and physical distancing: The need for prevention and early intervention," *JAMA internal medicine*, 2020, 180 (6), 817–818.
- **Gallup, I**, "Understanding how Gallup uses the Cantril Scale: Development of the 'Thriving, Struggling, Suffering'categories," *Retrieved January* 14, 2020, 2013.
- Giang, Kim Bao, Peter Allebeck, Gunnar Kullgren, and Nguyen Van Tuan, "The Vietnamese version of the Self Reporting Questionnaire 20 (SRQ-20) in detecting mental disorders in rural Vietnam: a validation study," *International Journal of Social Psychiatry*, 2006, 52 (2), 175–184.
- Giuntella, Osea, Kelly Hyde, Silvia Saccardo, and Sally Sadoff, "Lifestyle and Mental Health Disruptions During COVID-19," *Available at SSRN 3666985*, 2020.
- González-Sanguino, Clara, Berta Ausín, Miguel ÁngelCastellanos, Jesús Saiz, Aída López-Gómez, Carolina Ugidos, and Manuel Muñoz, "Mental health consequences during the initial stage of the 2020 Coronavirus pandemic (COVID-19) in Spain," *Brain*, *Behavior*, and Immunity, 2020.
- González-Sanguino, Clara, Berta Ausín, Miguel Ángel Castellanos, Jesús Saiz, Aída López-Gómez, Carolina Ugidos, and Manuel Muñoz, "Mental health consequences during the initial stage of the 2020 Coronavirus pandemic (COVID-19) in Spain," 2020, *87*, 172–176.

- Gupta, Sumedha, Laura Montenovo, Thuy D Nguyen, Felipe Lozano Rojas, Ian M Schmutte, Kosali I Simon, Bruce A Weinberg, and Coady Wing, "Effects of social distancing policy on labor market outcomes," Technical Report, National Bureau of Economic Research 2020.
- Harpham, Trudy, Michael Reichenheim, Rebecca Oser, Elizabeth Thomas, Narmeen Hamid, Surinder Jaswal, Ana Ludermir, and Magna Aidoo, "Measuring mental health in a cost-effective manner," *Health policy and planning*, 2003, 18 (3), 344–349.
- Haushofer, Johannes and Jeremy Shapiro, "The short-term impact of unconditional cash transfers to the poor: experimental evidence from Kenya," *The Quarterly Journal of Economics*, 2016, 131 (4), 1973–2042.
- Holman, E. Alison, Rebecca R. Thompson, Dana Rose Garfin, and Roxane Cohen Silver, "The unfolding COVID-19 pandemic: A probability-based, nationally representative study of mental health in the United States," *Science Advances*, 2020, 6 (42).
- Holmes, Emily A, Rory C O'Connor, V Hugh Perry, Irene Tracey, Simon Wessely, Louise Arseneault, Clive Ballard, Helen Christensen, Roxane Cohen Silver, Ian Everall et al., "Multidisciplinary research priorities for the COVID-19 pandemic: a call for action for mental health science," *The Lancet Psychiatry*, 2020.
- Holwerda, Tjalling J, Theo G van Tilburg, Dorly JH Deeg, Natasja Schutter, Rien Van, Jack Dekker, Max L Stek, Aartjan TF Beekman, and Robert A Schoevers, "Impact of loneliness and depression on mortality: results from the Longitudinal Ageing Study Amsterdam," *The British Journal of Psychiatry*, 2016, 209 (2), 127–134.
- Hwang, Tzung-Jeng, Kiran Rabheru, Carmelle Peisah, William Reichman, and Manabu Ikeda, "Loneliness and social isolation during the COVID-19 pandemic," *International psychogeriatrics*, 2020, 32 (10), 1217–1220.
- Iacoponi, Eduardo and Jair de Jesus Mari, "Reliability and factor structure of the Portuguese version of Self-Reporting Questionnaire," *International Journal of Social Psychiatry*, 1989, 35 (3), 213–222.
- **Idler, Ellen L and Yael Benyamini**, "Self-rated health and mortality: a review of twentyseven community studies," *Journal of health and social behavior*, 1997, pp. 21–37.
- Imbens, Guido and Karthik Kalyanaraman, "Optimal bandwidth choice for the regression discontinuity estimator," The Review of economic studies, 2012, 79 (3), 933–959.
- and Thomas Lemiuex, "Regression Discontinuity Designs: A Guide to Practice," Journal of Econometrics, 2008, 142 (2), 615–635.
- Jacobson, Nicholas C, Damien Lekkas, George Price, Michael V Heinz, Minkeun Song, A James O'Malley, and Paul J Barr, "Flattening the Mental Health Curve: COVID-19 Stay-at-Home Orders Are Associated With Alterations in Mental Health Search Behavior in the United States," JMIR mental health, 2020, 7 (6), e19347.
- Kessler, Ronald C, Gavin Andrews, Lisa J Colpe, Eva Hiripi, Daniel K Mroczek, S-LT Normand, Ellen E Walters, and Alan M Zaslavsky, "Short screening scales to monitor population prevalences and trends in non-specific psychological distress," *Psychological medicine*, 2002, 32 (6), 959–976.

- Kozlowski, Julian, Laura Veldkamp, and Venky Venkateswaran, "Scarring body and mind: the long-term belief-scarring effects of Covid-19," *NBER Working Paper*, 2020, (27439).
- Lee, Antoinette M, Josephine GWS Wong, Grainne M McAlonan, Vinci Cheung, Charlton Cheung, Pak C Sham, Chung-Ming Chu, Poon-Chuen Wong, Kenneth WT Tsang, and Siew E Chua, "Stress and psychological distress among SARS survivors 1 year after the outbreak," *The Canadian Journal of Psychiatry*, 2007, 52 (4), 233–240.
- Lee, David S. and David Card, "Regression discontinuity inference with specification error," *Journal of Econometrics*, 2008, 142 (2), 655–674.
- Lei, Lei, Xiaoming Huang, Shuai Zhang, Jinrong Yang, Lin Yang, and Min Xu, "Comparison of Prevalence and Associated Factors of Anxiety and Depression Among People Affected by versus People Unaffected by Quarantine During the COVID-19 Epidemic in Southwestern China," *Medical science monitor*, 2020, *26*, e924609–e924609.
- Leslie, Emily and Riley Wilson, "Sheltering in place and domestic violence: Evidence from calls for service during COVID-19," *forthcoming in Journal of Public Economics*, 2020.
- Liu, Nianqi, Fan Zhang, Cun Wei, Yanpu Jia, Zhilei Shang, Luna Sun, Lili Wu, Zhuoer Sun, Yaoguang Zhou, Yan Wang, and Weizhi Liu, "Prevalence and predictors of PTSS during COVID-19 outbreak in China hardest-hit areas: Gender differences matter," *Psychiatry research*, 2020, 287, 112921–112921.
- Lu, Haiyang, Peng Nie, and Long Qian, "Do quarantine experiences and attitudes towards COVID-19 affect the distribution of mental health in China? A quantile regression analysis," *Applied Research in Quality of Life*, 2020, pp. 1–18.
- **McCrary, Justin**, "Manipulation of the running variable in the regression discontinuity design: A density test," *Journal of econometrics*, 2008, 142 (2), 698–714.
- Moccia, Lorenzo, Delfina Janiri, Maria Pepe, Luigi Dattoli, Marzia Molinaro, Valentina De Martin, Daniela Chieffo, Luigi Janiri, Andrea Fiorillo, Gabriele Sani, and Marco Di Nicola, "Affective temperament, attachment style, and the psychological impact of the COVID-19 outbreak: an early report on the Italian general population," *Brain, behavior, and immunity*, 2020, 87, 75–79.
- Moghanibashi-Mansourieh, Amir, "Assessing the anxiety level of Iranian general population during COVID-19 outbreak," *Asian journal of psychiatry*, 2020, *51*, 102076–102076.
- Nelson, Daniel, "Violence against elderly people: a neglected problem," *The Lancet*, 2002, 360 (9339), 1094.
- Neria, Yuval, Arijit Nandi, and Sandro Galea, "Post-traumatic stress disorder following disasters: a systematic review," *Psychological medicine*, 2008, *38* (4), 467.
- **Newman, Michelle G and Nur Hani Zainal**, "The value of maintaining social connections for mental health in older people," *The Lancet Public Health*, 2020, 5 (1), e12–e13.
- **OECD**, "Pensions at a Glance 2017: Country Profiles Turkey," Technical Report, OECD 2017.

- Olagoke, Ayokunle A, Olakanmi O Olagoke, and Ashley M Hughes, "Exposure to coronavirus news on mainstream media: The role of risk perceptions and depression," *British journal of health psychology*, 2020, 25 (4), 865–874.
- Ortega, Francisco B, Duck chul Lee, Xuemei Sui, Laura D Kubzansky, Jonatan R Ruiz, Meghan Baruth, Manuel J Castillo, and Steven N Blair, "Psychological well-being, cardiorespiratory fitness, and long-term survival," *American journal of preventive medicine*, 2010, 39 (5), 440–448.
- **Ozamiz-Etxebarria, Naiara, Maria Dosil-Santamaria, Maitane Picaza-Gorrochategui, and Nahia Idoiaga-Mondragon**, "Stress, anxiety, and depression levels in the initial stage of the COVID-19 outbreak in a population sample in the northern Spain," *Cadernos de saúde pública*, 2020, *36* (4), e00054020–e00054020.
- Panchal, Nirmita, Rabah Kamal, Kendal Orgera, Cynthia Cox, Rachel Garfield, Liz Hamel, and P Chidambaram, "The implications of COVID-19 for mental health and substance use," *KFF Issue Brief*, 2020, *5*, 2020.
- Patel, V, R Araya, N Chowdhary, M King, B Kirkwood, S Nayak, G Simon, and HA Weiss, "Detecting common mental disorders in primary care in India: a comparison of five screening questionnaires," *Psychological medicine*, 2008, 38 (2), 221.
- **Pfefferbaum, Betty and Carol S North**, "Mental health and the Covid-19 pandemic," *New England Journal of Medicine*, 2020.
- **Proto, Eugenio and Climent Quintana-Domeque**, "COVID-19 and Mental Health Deterioration among BAME Groups in the UK," 2020.
- **Qiu, Jianyin, Bin Shen, Min Zhao, Zhen Wang, Bin Xie, and Yifeng Xu**, "A nationwide survey of psychological distress among Chinese people in the COVID-19 epidemic: implications and policy recommendations," *General psychiatry*, 2020, 33 (2), e100213–e100213.
- Ravindran, Saravana and Manisha Shah, "Unintended Consequences of Lockdowns: COVID-19 and the Shadow Pandemic," Technical Report, National Bureau of Economic Research 2020.
- **Romano, Joseph P and Michael Wolf**, "Efficient computation of adjusted p-values for resampling-based stepdown multiple testing," *Statistics & Probability Letters*, 2016, 113, 38–40.
- Samadarshi, Saurav Chandra Acharya, Sharmistha Sharma, and Jeevan Bhatta, "An online survey of factors associated with self-perceived stress during the initial stage of the COVID-19 outbreak in Nepal," *Ethiopian Journal of Health Development*, 2020, 34 (2).
- Samuel, Samantha Ciarocco Dotson and Katherine A Koh, "Disaster psychiatry and homelessness: creating a mental health COVID-19 response," *The Lancet Psychiatry*, 2020.
- Santini, Ziggi Ivan, Paul E Jose, Erin York Cornwell, Ai Koyanagi, Line Nielsen, Carsten Hinrichsen, Charlotte Meilstrup, Katrine R Madsen, and Vibeke Koushede, "Social disconnectedness, perceived isolation, and symptoms of depression and anxiety among older Americans (NSHAP): a longitudinal mediation analysis," *The Lancet Public Health*, 2020, *5* (1), e62–e70.

- Simes, R. J., "An improved Bonferroni procedure for multiple tests of significance," *Biometrika*, 1986, 73 (3), 751–754.
- Steptoe, Andrew, Aparna Shankar, Panayotes Demakakos, and Jane Wardle, "Social isolation, loneliness, and all-cause mortality in older men and women," *Proceedings of the National Academy of Sciences*, 2013, 110 (15), 5797–5801.
- Sønderskov, Kim Mannemar, Peter Thisted Dinesen, Ziggi Ivan Santini, and Søren Dinesen Østergaard, "The depressive state of Denmark during the COVID-19 pandemic," *Acta neuropsychiatrica*, 2020, 32 (4), 226–228.
- **The New York Times**, "The Coronavirus Mental Health Crisis Hits Home," *The New York Times*, May 12, 2020.
- **Tubadji, Annie, Frederic Boy, and Don J Webber**, "Narrative economics, public policy and mental health," *Covid Economics*, 2020, 20, 109–131.
- **Tufan, I**, "Violence against the elderly in Turkey, 1st Turkey atlas of Gerontology-Geroatlas," 2011.
- Tull, Matthew T, Keith A Edmonds, Kayla Scamaldo, Julia R Richmond, Jason P Rose, and Kim L Gratz, "Psychological Outcomes Associated with Stay-at-Home Orders and the Perceived Impact of COVID-19 on Daily Life," *Psychiatry research*, 2020, p. 113098.
- van der Westhuizen, Claire, Gail Wyatt, John K Williams, Dan J Stein, and Katherine Sorsdahl, "Validation of the self reporting questionnaire 20-item (SRQ-20) for use in a low-and middle-income country emergency centre setting," *International journal of mental health and addiction*, 2016, 14 (1), 37–48.
- Wang, Cuiyan, Riyu Pan, Xiaoyang Wan, Yilin Tan, Linkang Xu, Cyrus S Ho, and Roger C Ho, "Immediate psychological responses and associated factors during the initial stage of the 2019 coronavirus disease (COVID-19) epidemic among the general population in China," *International journal of environmental research and public health*, 2020, 17 (5), 1729.
- _ , _ , _ , _ , _ , _ , Roger S McIntyre, Faith N Choo, Bach Tran, Roger Ho, Vijay K Sharma et al., "A longitudinal study on the mental health of general population during the COVID-19 epidemic in China," *Brain, behavior, and immunity*, 2020.
- Wang, Huiyao, Qian Xia, Zhenzhen Xiong, Zhixiong Li, Weiyi Xiang, Yiwen Yuan, Yaya Liu, and Zhe Li, "The psychological distress and coping styles in the early stages of the 2019 coronavirus disease (COVID-19) epidemic in the general mainland Chinese population: A web-based survey," *Plos one*, 2020, *15* (5), e0233410.
- **WHO**, "A User's guide to the self reporting questionnaire (SRQ," Technical Report, World Health Organization 1994.
- World Health Organization, "The WHO special initiative for mental health (2019-2023): universal health coverage for mental health," Technical Report, World Health Organization 2019.
- Xiong, Jiaqi, Orly Lipsitz, Flora Nasri, Leanna MW Lui, Hartej Gill, Lee Phan, David Chen-Li, Michelle Iacobucci, Roger Ho, Amna Majeed et al., "Impact of COVID-19

pandemic on mental health in the general population: A systematic review," *Journal of affective disorders*, 2020.

- Yokoyama, Yukari, Kotaro Otsuka, Norito Kawakami, Seiichiro Kobayashi, Akira Ogawa, Kozo Tannno, Toshiyuki Onoda, Yumi Yaegashi, and Kiyomi Sakata, "Mental health and related factors after the Great East Japan earthquake and tsunami," *PloS one*, 2014, *9* (7), e102497.
- Zhang, Yingfei and Zheng Feei Ma, "Impact of the COVID-19 pandemic on mental health and quality of life among local residents in Liaoning Province, China: A cross-sectional study," *International journal of environmental research and public health*, 2020, 17 (7), 2381.
- Özdin, Selçuk and Şükriye Bayrak Özdin, "Levels and predictors of anxiety, depression and health anxiety during COVID-19 pandemic in Turkish society: The importance of gender," *International journal of social psychiatry*, 2020, *66* (5), 504–511.





Note: The sample includes all individuals born before and after 44 months around the cutoff point, December 1955. The figures plot predetermined covariates in monthly bins against the month-year of birth of being born in December 1955. The vertical line in each graph represents the cutoff point. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B.





the cutoff point, December 1955. The vertical line in each graph represents the cut-off point, December 1955. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B. Note: The figures plot the non-residualized values of the number of days spent outside last week, the probability of being subject to the curfew, and the probability of never going outside against the month-year of birth of being born in December 1955. The sample includes all individuals born before and after 44 months around



Note: The figures plot the residualized values (after controlling for all variables in the main specification other than distance to the cutoff) of the indices of mental distress outcomes over the month-year of birth of being born in December 1955. The sample includes all individuals born before and after 44 months around the cutoff point, December 1955. The vertical line in each graph represents the cut-off point, December 1955. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B.

	Mean	S.D.	Min	Max	Obs.
Panel A: Pre-determined Characteristics					
Completed high school	0.27	0.44	0	1	1896
Illiterate	0.13	0.34	0	1	1896
Female	0.43	0.50	0	1	1909
Married	0.81	0.40	0	1	1907
Widowed or separated	0.18	0.38	0	1	1907
Non-Turkish	0.26	0.44	0	1	1881
Pre-Covid-19 household size	3.29	1.73	1	10	1909
Ever received psychological support	0.11	0.31	0	1	1887
Has a chronic disease	0.57	0.50	0	1	1898
Panel B: Mobility Outcomes					
Days outside last week	1.87	1.79	0	7	1896
Under curfew	0.48	0.50	0	1	1907
Never goes out	0.36	0.48	0	1	1885
Panel C: Potential Channels					
Paid employed	0.13	0.34	0	1	1892
Paid or unpaid employed	0.15	0.36	0	1	1897
Has a job but could not attend last week	0.14	0.35	0	1	1876
Has money for usual needs	0.61	0.49	0	1	1894
Worried about spending money	0.60	0.49	0	1	1890
Limited social interaction	0.60	0.49	0	1	1905
Limited physical activity	0.55	0.50	0	1	1878
Current household size	3.42	1.83	1	10	1909
Conflict with a household member	0.37	0.48	0	1	1868

TABLE 1: SUMMARY STATISTICS FOR 59-70 YEAR-OLD INDIVIDUALS

Notes: The table presents the means, standard deviations, minimum values, maximum values, and number of observations. The sample includes 59–70 year-old individuals born within 72 months before and after December 1955. The variables are described in Appendix B.

	Mean	S.D.	Min	Max	Obs.
Over the last 4 weeks,					
have you often had headaches?	0.47	0.50	0	1	1904
has your appetite been poor?	0.38	0.49	0	1	1903
have you slept badly?	0.49	0.50	0	1	1904
have you been easily frightened?	0.31	0.46	0	1	1897
have you had shaking hands?	0.19	0.39	0	1	1894
have you felt nervous, tense, or worried?	0.66	0.47	0	1	1891
has your digestion been poor?	0.32	0.47	0	1	1890
have you had trouble in thinking clearly?	0.41	0.49	0	1	1889
have you felt unhappy?	0.57	0.50	0	1	1890
have you cried more often than usual?	0.18	0.38	0	1	1877
have you found it difficult to enjoy your daily activities?	0.52	0.50	0	1	1887
have you found it difficult to make decisions?	0.36	0.48	0	1	1888
has your daily work suffered?	0.46	0.50	0	1	1887
have you been unable to play a useful part in life?	0.22	0.41	0	1	1883
have you lost interest in things?	0.38	0.49	0	1	1880
have you felt that you are a worthless person?	0.21	0.41	0	1	1879
has the thought of ending your life been on your mind?	0.04	0.20	0	1	1882
have you felt tired all the time?	0.50	0.50	0	1	1895
have you had uncomfortable feelings in your stomach?	0.42	0.49	0	1	1899
have you gotten tired easily?	0.56	0.50	0	1	1895

TABLE 2: LIST OF SRQ-20 QUESTIONS AND THEIR SUMMARY STATISTICS

Notes: The table presents the list of questions included in the SRQ-20 inventory, and their summary statistics, including the means, standard deviations, minimum values, maximum values, and number of observations. The sample includes 59–70 year-old individuals born within 72 months before and after December 1955. The variables are described in Appendix B.

	±17	±30	±45	±60
Days outside last week				
Born before 1955	-1.011	-1.106	-1.09	-1.023
	(0.308)	(0.230)	(0.180)	(0.166)
	[0.002]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.003 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$
Observations	506	832	1214	1601
Control group mean	2.30	2.33	2.30	2.40
Under curfew				
Born before 1955	0.609	0.663	0.708	0.723
	(0.071)	(0.057)	(0.045)	(0.036)
	[<0.001]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$
Observations	511	837	1222	1610
Control group mean	0.12	0.09	0.09	0.08
Never goes out				
Born before 1955	0.213	0.301	0.297	0.245
	(0.079)	(0.045)	(0.036)	(0.034)
	[0.011]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.004 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$
Observations	500	825	1203	1591
Control group mean	0.17	0.18	0.20	0.18

TABLE 3: EFFECTS OF CURFEW ON MOBILITY OUTCOMES

Notes: This table presents first-stage estimates of the effect of being born before December 1955 on the mobility outcomes of individuals. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding p-values and Anderson (2008)'s sharpened q-values are in square and angle brackets, respectively.

	±17	±30	±45	±60
Mental distress index				
Born before 1955	0.264	0.359	0.205	0.238
	(0.188)	(0.115)	(0.094)	(0.077)
	[0.169]	[0.003]	[0.032]	[0.003]
	$\langle 0.422 \rangle$	$\langle 0.011 \rangle$	$\langle 0.095 \rangle$	$\langle 0.011 \rangle$
Observations	475	777	1133	1485
Somatic symptoms of distress i	ndex			
Born before 1955	0.343	0.296	0.175	0.198
	(0.195)	(0.121)	(0.085)	(0.080)
	[0.087]	[0.018]	[0.043]	[0.015]
	$\langle 0.422 \rangle$	$\langle 0.019 \rangle$	$\langle 0.095 \rangle$	$\langle 0.016 \rangle$
Observations	503	824	1203	1580
Nonsomatic symptoms of distre	ess index			
Born before 1955	0.153	0.272	0.158	0.188
	(0.182)	(0.118)	(0.092)	(0.073)
	[0.404]	[0.025]	[0.088]	[0.011]
	$\langle 0.422 \rangle$	$\langle 0.019 \rangle$	$\langle 0.095 \rangle$	$\langle 0.016 \rangle$
Observations	478	781	1137	1491
Sum of "yes" answers in SRQ-	-20			
Born before 1955	1.163	1.236	0.734	0.816
	(0.936)	(0.517)	(0.440)	(0.384)
	[0.222]	[0.02]	[0.098]	[0.036]
	$\langle 0.422 \rangle$	$\langle 0.019 \rangle$	$\langle 0.095 \rangle$	$\langle 0.021 \rangle$
Observations	475	777	1133	1485
Control group mean	7.03	7.17	7.05	7.16

TABLE 4: EFFECTS OF CURFEW ON MENTAL HEALTH OUTCOMES

Notes: This table presents regression discontinuity estimates of the effect of the curfew on mental health outcomes. See the Appendix B for details of index construction. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±17	±30	±45	±60
Panel A: Employment and Ir	come Outcomes			
Paid employed				
Born before 1955	-0.032	-0.083	-0.071	-0.064
	(0.074)	(0.048)	(0.037)	(0.032)
	[0.671]	[0.090]	[0.059]	[0.044]
	(1.000)	(0.812)	(0.423)	(0.285)
Observations	506	832	1215	1597
Control group mean	0.16	0.15	0.16	0.18
Paid or unpaid employed				
Born before 1955	-0.020	-0.031	-0.025	-0.039
	(0.083)	(0.052)	(0.041)	(0.034)
	[0.814]	[0.549]	[0.542]	[0.263]
	(1.000)	$\langle 1.000 \rangle$	(1.000)	(0.358)
Observations	506	832	1216	1601
Control group mean	0.18	0.18	0.19	0.21
Has a job but could not attend la	ist week			
Born before 1955	-0.016	-0.010	-0.022	-0.042
	(0.090)	(0.047)	(0.035)	(0.032)
	[0.864]	[0.836]	[0.527]	[0.187]
	(1.000)	(1.000)	(1.000)	(0.333)
Observations	500	825	1207	1586
Control group mean	0.14	0.16	0.16	0.18
Has enough money for usual new	eds			
Born before 1955	0.028	0.044	0.063	0.074
	(0.096)	(0.062)	(0.058)	(0.048)
	[0.775]	[0.481]	[0.276]	[0.121]
	(1.000)	(1.000)	(1.000)	(0.320)
Observations	507	832	1218	1601
Control group mean	0.58	0.56	0.58	0.57
Worried about spending money				
Born before 1955	-0.120	-0.045	-0.021	-0.030
	(0.066)	(0.052)	(0.047)	(0.042)
	[0.080]	[0.393]	[0.660]	[0.478]
	$\langle 0.670 \rangle$	(1.000)	(1.000)	(0.434)
Observations	506	829	1213	1597
Control group mean	0.62	0.59	0.60	0.62

TABLE 5: EFFECTS OF CURFEW ON POTENTIAL CHANNELS

	±17	±30	±45	±60
Panel B: Social and Physical	Isolation Outcom	es		
Limited social interaction				
Born before 1955	0.222	0.074	0.088	0.083
	(0.071)	(0.058)	(0.051)	(0.042)
	[0.004]	[0.211]	[0.088]	[0.049]
	$\langle 0.002 \rangle$	$\langle 0.119 \rangle$	$\langle 0.047 \rangle$	$\langle 0.026 \rangle$
Observations	509	835	1220	1608
Control group mean	0.61	0.57	0.56	0.53
Limited physical activity				
Born before 1955	0.365	0.203	0.248	0.219
	(0.092)	(0.064)	(0.054)	(0.047)
	[<0.001]	[0.003]	[<0.001]	[<0.001]
	$\langle 0.001 \rangle$	$\langle 0.006 \rangle$	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$
Observations	499	822	1201	1585
Control group mean	0.45	0.45	0.46	0.45
Panel C: Household Conflict	Outcomes			
Household size				
Born before 1955	-0.098	0.032	-0.047	0.013
	(0.244)	(0.183)	(0.161)	(0.146)
	[0.690]	[0.860]	[0.771]	[0.930]
	$\langle 1.000 \rangle$	$\langle 0.756 \rangle$	$\langle 0.784 \rangle$	$\langle 1.000 \rangle$
Observations	511	838	1224	1612
Control group mean	3.53	3.50	3.50	3.54
Conflict with a household membe	er			
Born before 1955	0.035	0.091	0.052	0.023
	(0.093)	(0.055)	(0.042)	(0.037)
	[0.705]	[0.105]	[0.220]	[0.523]
	$\langle 1.000 \rangle$	$\langle 0.265 \rangle$	$\langle 0.784 \rangle$	$\langle 1.000 \rangle$
Observations	498	820	1200	1579
Control group mean	0.34	0.36	0.38	0.38

TABLE 5: EFFECTS OF CURFEW ON POTENTIAL CHANNELS, CONT'D.

Notes: This table presents regression discontinuity estimates of the effect of the curfew on potential channels. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±17	±30	±45	±60
Supports the 65+ age-specific curfew				
Born before 1955	-0.222	-0.129	-0.115	-0.086
	(0.088)	(0.049)	(0.041)	(0.034)
	[0.017]	[0.010]	[0.006]	[0.014]
	(0.036)	$\langle 0.022 \rangle$	$\langle 0.012 \rangle$	(0.029)
Observations	505	828	1207	1590
Control group mean	0.83	0.80	0.79	0.81
Satisfied with the government's Covid-19 policy response				
Born before 1955	-0.122	-0.093	-0.032	-0.002
	(0.099)	(0.051)	(0.047)	(0.038)
	[0.227]	[0.071]	[0.490]	[0.958]
	$\langle 0.129 \rangle$	$\langle 0.037 \rangle$	(0.325)	$\langle 0.919 \rangle$
Observations	501	822	1197	1579
Control group mean	0.74	0.70	0.69	0.69

TABLE 6: EFFECTS OF CURFEW ON POLITICAL SUPPORT FOR CURFEW AND COVID-19-SPECIFIC POLICIES

Notes: This table presents regression discontinuity estimates of the effect of the curfew on supporting the 65+ agespecific curfew and being satisfied with the government's Covid-19 policy response using a linear control function. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

Study	Empirical strategy	Sample	Main finding
Adams-Prassl et al. (2020)	Difference-in- differences methodology.	Two-waves of repeated cross- section data collected in March and April 2020 from employed adults who live in the U.S.	Mental health score of individuals who live in the states that imposed a COVID-19 lockdown is 0.85SD below below compared to those who did not.
Armbruster and Klotzbücher (2020)	Event study design.	Daily contacts to 91 healthline-centers in Germany collected between 01/01/2019 and 04/28/2020.	20% rise in counsel- ing requests during the week of lock- down, even stronger increase in areas with stricter measures.
Banks and Xu (2020)	Difference between observed and coun- terfactual outcomes in the absence of pandemic in April 2020. Counterfactual outcomes are pre- dicted by regression models using past data, including indi- vidual fixed-effects.	UK Household Longitudinal Study (UKHLS) waves 1-9 (2009-2019) merged with April 2020 COVID-19 Survey.	Mental health of adults in the U.K. deteriorated by 8.1% due to Covid-19 pandemic, with larger drops among younger and female population.
Beland et al. (2020)	Comparison of pre- and post-COVID out- comes adjusted for pre-determined covariates.	Canadian Perspective Survey Series 1 - Impacts of COVID-19 (CPSS)	Workers who are most severely affected by the pandemic report worse self-reported mental health than their peers.
Brodeur et al. (2020)	Difference-in- differences methodology and Regression Dis- continuity Design.	Google Trends data related to 13 pre-defined well-being terms between January 1 st - April 10 th (2019 and 2020) from Europe and the U.S.	Following the re- stricted mobility, search intensity of Google users in- crease for boredom, loneliness, worry and decrease for stress, suicide, and divorce.

Appendix A A review of the literature on the effects of Covid-19 on mental health

Burdett et al. (2020)	Difference-in-	Covid-19 module	During the lock-
	differences method-	from the UK House-	downs, weather
	ology that rely on	hold Longitudinal	patterns (tempera-
	variation in local	Study (UKHLS)	ture, sunshine, and
	weather conditions.	April, May, June, and	rainfall) have very lit-
		July 2020 merged	tle differential impact
		with the waves	on mental health
		10-11 (2019) of	despite its strong
		the main survey.	impact on mobility.
Daly et al. (2020)	Longitudinal	Covid-19 module	Proportion of
	trend analysis.	from the UK House-	individuals who
		hold Longitudinal	report mental health
		Study (UKHLS)	problems increased
		April, May, and June	by 13.5 percentage
		2020 merged with the	points from 2017-
		wave 9 (2017-2019)	2019 baseline to April
		of the main survey.	2020 and remained
			high until June 2020.
Etheridge and	Longitudinal trend	UK Household	The decrease in
Spantig (2020)	analysis by gender.	Longitudinal Study	mental health
		(UKHLS) waves 1-9	among the UK
		(2009-2019) merged	population is more
		with April 2020	than twice larger
		COVID-19 Survey.	among women.
			Social factors such as
			loneliness explains
			the gender gap.
Fetzer et al. (2020a)	Trend analysis and	Google Trends	Economic anxiety
	online random-	data related to	exhibits a strong
	ized experiment.	economic anxiety	upward trend in
		("Recession", "Stock	parallel to the course
		Market Crash",	of the pandemic.
		"Conspiracy Theory",	Access to informa-
		"Survivalism"), two	tion and ways of
		online experimental	communicating
		surveys from the	directly affect the
		U.S. (March 5^{th} and	economic anxiety.
		March 16^{th} , 2020.	
Fetzer et al. (2020b)	Descriptive anal-	Survey data collected	Strong government
	ysis and event	from 100,000	response to COVID-
	study design.	participants and 58	19 leads to a decrease
		countries in March	in the likelihood
		and early April 2020.	of respondents
			to report worry
			and depression.

Giuntella et al. (2020)	Longitudinal analysis	Wearable health	Substantial decrease
	of survey and	device data from	in physical activity,
	biometric data.	University of	increase in phone
		Pitsburgh students	interaction and sleep,
		between Spring 2019	and 65% increase
		and Spring 2020	in depression risk.
		with baseline and	
		end-line surveys	
		in each semester.	
Holman et al. (2020)	Longitudinal analysis	U.S. probability-	Increased likelihood
	of survey data	based nationally	of reporting symp-
		representative survey	toms of acute stress
		NORC AmeriSpeak	and depression. Poor
		panel, three waves	baseline health and
		collected between	media exposure
		March 18 th and	further deteriorates
		April 18 th , 2020.	mental health.
Proto and Quintana-	Longitudinal	Covid-19 module	Black, Asian, and
Domeque (2020)	subgroup analysis	from the UK House-	other minorities in
	of survey data	hold Longitudinal	the UK experience
		Study (UKHLS)	a larger decrease
		April 2020 merged	in mental health
		with the wave	compared to the
		9 (2017-2019) of	white population.
		the main survey.	
Tubadji et al. (2020)	Difference-in-	Google Trends data	Lockdowns have
	differences	related to mental	a negative impact
	methodology	health ("death",	on mental health
	and Regression Dis-	"suicide") collected	through experienced
	continuity Design.	from Italy and UK	fear of death.
		collected between	
		March 12^{th} and	
		March 23 rd , 2020.	

We also note that our results add to the documented adverse mental health impacts following large-scale natural disasters and stressful events such as Zika and SARS outbreaks, major earthquakes, and terrorist attacks.^{43,44} In addition, our paper also contributes to the growing literature on the effects of pandemic-driven social isolation on at-risk populations, including adolescents, elderly people, homeless people, people with disabilities, and people with mental health concerns (Pfefferbaum and North 2020; Dotson and Koh 2020; Armitage and Nellums 2020). Given their heightened risk of physical and mental health problems, exposure to social isolation is a particularly important concern for the older adults we study in this paper. However, since several other high-risk groups also face the risk of adverse mental consequences due to social isolation, our findings have broader implications for evaluating the risks for such groups.⁴⁵ Finally, potential scarring effects could impact the long-term mental health of isolated individuals; these effects are likely to pose problems long after the stay-at-home orders cease.⁴⁶

⁴³See, for example, Galea et al. (2002), Lee et al. (2007), Neria et al. (2008), Yokoyama et al. (2014), and Galea et al. (2020).

⁴⁴Rapidly growing literature on the impact of COVID-19 on mental wellbeing faces similar identification challenges. The majority of these studies report increased symptoms or cross-sectional correlates of mental health problems experienced by the general population of countries affected by the pandemic. These studies include, but are not limited to, Ahmed et al. (2020); González-Sanguino et al. (2020); Hwang et al. (2020); Lei et al. (2020); Liu et al. (2020); Lu et al. (2020); Moccia et al. (2020); Moghanibashi-Mansourieh (2020); Olagoke et al. (2020); Ozamiz-Etxebarria et al. (2020); Özdin and Bayrak Özdin (2020); El-Zoghby et al. (2020); Qiu et al. (2020); Samadarshi et al. (2020); Sønderskov et al. (2020); Wang et al. (2020a,b,c); Zhang and Ma (2020). See Xiong et al. (2020) for a more in-depth discussion of this literature.

⁴⁵These risks are particularly serious for children and adolescents with special needs or disadvantages, such as disabilities, trauma experiences, and existing mental health problems (Fegert et al. 2020).

⁴⁶While previous studies have discussed the pandemic's scarring effects on long-term beliefs affecting economic outcomes (Kozlowski et al. 2020), scant attention has been given to the potential scarring effects on long-term mental health outcomes. When we consider the historical accounts of the Spanish flu, demographic evidence suggests that exposed populations reported depression, mental distraction, and sleep disturbances even six years after the pandemic (Eghigian 2020).

Appendix B List of Variables

Outcome Variables:

- Days outside last week: The number of days the respondent went outside last week.
- Under curfew: A dummy variable equal to one if the respondent reported being subject to the curfew within the last month.
- Never goes out: A dummy variable equal to one if the respondent reported his/her current frequency of going outside as "never going outside".
- Mental distress indices: Following Anderson (2008), each index is generated by demeaning its component outcomes and converting them to effect sizes through dividing by control group standard deviation. The demeaned values are subsequently combined by weighting according to the inverse of the covariance matrix.
 - Somatic symptoms of distress index: a z-score calculated by averaging the z-scores from each of the 4 somatic symptoms of distress indicators, including dummy variables equal to one if the respondent reports that she experienced the following within the last four weeks: (i) frequent headaches, (ii) shaking hands, (iii) poor digestion, and (iv) uncomfortable feelings in the stomach.
 - Nonsomatic symptoms of distress index: a z-score calculated by averaging the z-scores from each of the 16 nonsomatic symptoms of distress indicators, including dummy variables equal to one if the respondent reports that she experienced the following within the last four weeks: (i) poor appetite, (ii) sleeping badly, (iii) been easily frightened, (iv) felt nervous, tense, or worried, (v) had trouble in thinking clearly, (vi) felt unhappy, (vii) cried more often than usual, (viii) found it difficult to enjoy daily activities, (ix) found it difficult to make decisions, (x) daily work suffered, (xi) been unable to play a useful part in life, (xii) lost interest in things, (xiii) felt that he/she was a worthless person, (xiv) thought about suicide, (xv) felt tired all the time, and (xvi) got tired easily.
 - Mental distress index: A z-score calculated by averaging the z-scores from 20 symptoms
 of mental distress indicators, including 4 somatic and 16 nonsomatic indicators, as listed
 above.
- Paid employed: A dummy variable equal to one if the respondent reported working to earn income in cash or kind in the reference week.
- Paid or unpaid employed: A dummy variable equal to one if the respondent reported working to earn income or working as an unpaid family worker in the reference week.
- Has a job but could not attend last week: A dummy variable equal to one if the respondent reported having a job but could not attend this job last week.

- Has enough money for usual needs: A dummy variable equal to one if the respondent reported having enough money for satisfying his/her usual needs last month.
- Worried about spending money: A dummy variable equal to one if the respondent reported being worried about spending money last month.
- Limited social interaction: A dummy variable equal to one if the respondent reported that his/her social interaction with friends and family has been extremely limited or very limited in the last month compared to pre-Covid times.
- Limited physical activity: A dummy variable equal to one if the respondent reported that his/her physical activity (e.g. walking, running, doing sports, etc.) has been extremely limited or very limited in the last month compared to pre-Covid times.
- Household size: The number of people currently residing with the respondent in the same household.
- Conflict with a household member: A dummy variable equal to one if the respondent reported that he/she had a conflict with a household member last month.
- Supports the 65+ age-specific curfew: A dummy variable equal to one if the respondent reported being somewhat, very, or extremely supportive of the curfew policy.
- Satisfied with the government's Covid-19 policy response: A dummy variable equal to one if the respondent reported being somewhat, very, or extremely supportive of the government's policy response to Covid-19.

Covariates:

- Completed high school: A dummy variable equal to one if the respondent completed high school or above.
- Illiterate: A dummy variable equal to one if the respondent is illiterate.
- Female: A dummy variable equal to one if the respondent is female.
- Married: A dummy variable equal to one if the respondent is married.
- Widowed or separated: A dummy variable equal to one if the respondent is widowed or separated.
- Non-Turkish: A dummy variable equal to one if the respondent has a non-Turkish ethnic identity, e.g. Arabic, Kurdish, or other.
- Pre-Covid-19 household size: The number of people residing with the respondent in the same household prior to the Covid-19 outbreak.

- Ever received psychological support: A dummy variable equal to one if the respondent has ever received psychological support.
- Has a chronic disease: A dummy variable equal to one if the respondent has a chronic disease.

Outcome Variables in Appendix B:

- Poor physical health: A dummy variable equal to one of the respondent reports having a poor or very poor physical health.
- Poor mental health: A dummy variable equal to one of the respondent reports having a poor or very poor mental health.
- Suffering: A dummy variable equal to one if the respondent reports poor ratings of their current life situation (4 and below) and negative ratings for the next five years (4 and below).
- Each one of the below outcomes is a dummy variable that equals one if the respondent agreed with the statement:
 - *Considers himself/herself religious*: "Religion has an important place in my life."
 - Prays daily: "I prayed most of the day during the last month."
 - Agrees that one should live by the holy book: "One should live word-by-word the holy book."
 - Agrees that virus is a God-sent warning: "Epidemics is a God sent warning to humanity."
- Religiosity index: A standard normalized z-score calculated by averaging the individual 4 religiosity indicators defined above. Following Anderson (2008), the index is generated by demeaning its component outcomes and converting them to effect sizes through dividing by control group standard deviation. The demeaned values are subsequently combined by weighting according to the inverse of the covariance matrix.

Appendix C Survey Questionnaire

- 1. Province where the respondent lives in.
- 2. Type of residence
 - a. Rural
 - b. Urban
- 3. How many people are currently living in your household?
- 4. In normal times (prior to the pandemic), how many people live in your household?
- 5. What is your relationship to the household head?
 - a. Household head
 - b. Spouse
 - c. His/her children
 - d. Father / Mother
 - e. Brother / Sister
 - f. Father in law / Mother in law
 - g. Son in law / Daughter in law
 - h. Grandchild
 - i. Other relatives
 - j. Non-relatives
 - k. Housekeeper staying at home
- 6. Respondent's gender
 - a. Male
 - b. Female
- 7. How old are you?
- 8. What is your marital status?
 - a. Never married
 - b. Married
 - c. Divorced
 - d. Widowed

- 9. If married, how old is your spouse?
- 10. What is your education level, i.e. the highest degree of education you completed?
 - a. Illiterate.
 - b. Literate but not completed any educational institution.
 - c. Completed primary school (5 years of schooling)
 - d. Completed lower secondary, vocational and technical secondary school, or primary education
 - e. Completed upper secondary school (high school)
 - f. Completed 2- or 3-year higher education or faculty or 4 years higher education or faculty (university)
 - g. Completed Master's degree (5 or 6 years faculty included) or PhD
- 11. What is your year of birth as written in your national ID card? [Please ask the respondent to look at his/her ID card and tell.]
- 12. What is your month of birth as written in your national ID card? [Please ask the respondent to look at his/her ID card and tell.]
- 13. As you know there has been some recent regulations regarding going outside due to the coronavirus outbreak. Were you subject to the curfew for citizens 65 and plus over the last month? (Yes/No)
- 14. Over the last week, how many times did you go out?
- 15. Over the last month, how many times did you go out in a week on average?
- 16. As you know, those 65 and older were permitted to go outside in certain days of the week. Apart from these permits, how many times did you go out in the last week?
- 17. Which of the following explains your status regarding going outside?
 - a. I go out as much as I used to.
 - b. I go out less often.
 - c. I go out only to satisfy basic needs such as shopping.
 - d. I go out only to work.
 - e. I never go out.
- 18. Did you work to earn income in cash or kind in the reference week? (Yes/No)

- 19. (Ask if answer to 18 is NO) Did you work for an hour in the reference week in order to earn income or as unpaid family workers, even if you are a housewife, student or retired? (Yes/No).
- 20. Do you have a business or job in which you were temporarily absent in the reference week? (Yes/No)
- 21. (Ask if answer to 20 is YES) Why were you absent from this work in the reference week?
 - a. His/her illness, injury or temporary ill
 - b. Workplace shut down for economic reasons
 - c. Furlough
 - d. Government employment ban due to COVID-19
 - e. Laid off even though workplace did not shut down
 - f. Nature of work
 - g. There was no work
 - h. Other
- 22. What was your employment status at your most recent (or current) job?
 - a. Wage or salaried employee or casual workers
 - b. Employer
 - c. Self-employed
 - d. Unpaid family worker
- 23. What is the sector that you work in?
- 24. What was your occupation at your most recent (or current) job?
- 25. Do you have enough money to satisfy your usual needs compared to those times prior to the outbreak of COVID-19 crisis? (Yes/No)
- 26. In the last month, have you ever worried about spending money? (Yes/No)
- 27. Please imagine a ladder with steps numbered from zero at the bottom to ten at the top. The top of the ladder represents the best possible life for you and the bottom of the ladder represents the worst possible life for you. On which step of the ladder would you say you personally feel you stand at this time? (Scale:1-10)
- 28. On which step do you think you will stand about one year from now? (Scale:1-10)
- 29. Would you say your own health, in general, is

- a. Very good
- b. Good
- c. Fair
- d. Poor
- e. Very poor
- 30. Would you say your own physical health, is
 - a. Very good
 - b. Good
 - c. Fair
 - d. Poor
 - e. Very poor
- 31. Would you say your own mental health, is
 - a. Very good
 - b. Good
 - c. Fair
 - d. Poor
 - e. Very poor
- 32. Over the last month, how limited has your physical activity been compared to pre-Covid times?
 - a. Extremely limited
 - b. Very limited
 - c. Somewhat limited
 - d. Not so limited
 - e. Not at all limited.
- 33. Over the last month, how limited has your social interaction with your family/friend been compared to pre-Covid times?
 - a. Extremely limited
 - b. Very limited
 - c. Somewhat limited
 - d. Not so limited

e. Not at all limited.

- 34. Over the last 4 weeks, have you often had headaches? (Yes/No)
- 35. Over the last 4 weeks, has your appetite been poor? (Yes/No)
- 36. Over the last 4 weeks, have you slept badly? (Yes/No)
- 37. Over the last 4 weeks, have you been easily frightened? (Yes/No)
- 38. Over the last 4 weeks, have you had shaking hands? (Yes/No)
- 39. Over the last 4 weeks, have you felt nervous, tense, or worried? (Yes/No)
- 40. Over the last 4 weeks, has your digestion been poor? (Yes/No)
- 41. Over the last 4 weeks, have you had trouble in thinking clearly? (Yes/No)
- 42. Over the last 4 weeks, have you cried more often than usual? (Yes/No)
- 43. Over the last 4 weeks, have you found it difficult to enjoy your daily activities? (Yes/No)
- 44. Over the last 4 weeks, have you found it difficult to make decisions? (Yes/No)
- 45. Over the last 4 weeks, has your daily work suffered? (Yes/No)
- 46. Over the last 4 weeks, have you been unable to play a useful part in life? (Yes/No)
- 47. Over the last 4 weeks, have you lost interest in things? (Yes/No)
- 48. Over the last 4 weeks, have you felt that you are a worthless person? (Yes/No)
- 49. Over the last 4 weeks, has the thought of ending your life been on your mind? (Yes/No)
- 50. Over the last 4 weeks, have you felt tired all the time? (Yes/No)
- 51. Over the last 4 weeks, have you had uncomfortable feelings in your stomach? (Yes/No)
- 52. Over the last 4 weeks, have you gotten tired easily? (Yes/No)
- 53. Over the last 4 weeks, have you had high blood pressure? (Yes/No)
- 54. Do you have a chronic disease for which you regularly take medication, such as high blood pressure, diabetes, or heart disease?
- 55. Have you ever visited an expert to receive psychological treatment before the lockdown began? (Yes/No)
- 56. Since the lockdown began, have you seen a healthcare professional? (Yes/No)

- 57. Over the last 4 weeks, how many cigarettes have you smoked on average per day?
- 58. How satisfied are you from the government's policy response to Covid-19?
 - a. Extremely satisfied
 - b. Very satisfied
 - c. Somewhat satisfied
 - d. Not so satisfied
 - e. Not at all satisfied.
- 59. How much do you support the curfew for individuals 65 and older?
 - a. Extremely supportive
 - b. Very supportive
 - c. Somewhat supportive
 - d. Not so supportive
 - e. Not at all supportive.
- 60. Over the last month, did you experience a conflict with one of the household members?
- 61. If yes, which household members did you experience the conflict with? (Choose as many as applicable.)
 - a. Spouse
 - b. His/her children
 - c. Father / Mother
 - d. Brother / Sister
 - e. Father in law / Mother in law
 - f. Son in law / Daughter in law
 - g. Grandchild
 - h. Other relatives
 - i. Non-relatives
 - j. Housekeeper staying at home
- 62. Do you agree with the following statements?
 - a. Religion has an important place in my life.
 - b. I prayed most of the day during the last month.

- c. One should live word-by-word the holy book.
- d. Epidemics is a God-sent warning to humanity.
- 63. We are all citizens of the Republic of Turkey, but we may have different ethnic backgrounds. How do you define your ethnic identity?
 - a. Turkish
 - b. Kurdish
 - c. Arab
 - d. Other
- 64. What is the total monthly income of all household members? Including all income earned by every household member, how much is the average sum of earnings in a month?

Appendix D Additional Figures and Tables



Figure A1: Retirement by AGE: Household Labor Force Survey 2019

Note: The sample includes all individuals born between January 1950 and December 1961. The vertical line represents the cut-off point by age in 2019.



Note: The sample includes all individuals born between January 1950 and December 1961. The vertical line represents the birth year and birth month for which the estimated coefficient of difference in exposure to curfew between the treatment and the control group is maximum. Variable definitions are listed in Appendix B.

FIGURE A3: NONRESPONSE RATE AROUND THE AGE THRESHOLD



Note: The figure plots the nonresponse rate against the age of the respondent centered on 65. The sample includes all individuals born before and after 5 years around the cutoff age of 65. The vertical line in each graph represents the cut-off point, age 65. Gray lines show 95 percent confidence intervals around the mean level. The outcome variable captures the nonresponse rate of individuals who either did not answer the call, or refused to answer survey questions.



Note: The sample includes all individuals born between January 1950 and December 1961. The vertical line in each graph represents the cut-off point, December 1955. Circles indicate the raw number of observations for each birth month-year bin. Gray lines show 95 percent confidence intervals around the quadratic local polynomial. Variable definitions are listed in Appendix B.



FIGURE A5: RD TREATMENT EFFECTS ON MOBILITY OUTCOMES USING RESIDUALIZED OUTCOMES

Note: The figures plot the residualized values (after controlling for all variables in the main specification other than distance to the cutoff) of the number of days spent outside last week, the probability of being subject to the curfew, and the probability of never going outside against the month-year of birth of being born in December 1955. The sample includes all individuals born before and after 44 months around the cutoff point, December 1955. The vertical line in each graph represents the cut-off point, December 1955. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B.



Figure A6: RD Treatment Effects on Mental Health Outcomes using Non-residualized Outcomes

Note: The figures plot the non-residualized values of the indices of mental distress outcomes over the month-year of birth of being born in December 1955. The sample includes all individuals born before and after 44 months around the cutoff point, December 1955. The vertical line in each graph represents the cut-off point, December 1955. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B.





Note: The figures plot the residualized values of potential channel outcomes over the month-year of birth of being born in December 1955. The sample includes all individuals born before and after 44 months around the cutoff point, December 1955. The vertical line in each graph represents the cut-off point, December 1955. Gray lines show 95 percent confidence intervals around the mean level. Variable definitions are listed in Appendix B.

4

22 nths)

-22 0 Born before December 1955 (in mo

44

4

-22 0 22 Born before December 1955 (in months)

4

4

-22 0 22 Born before December 1955 (in months)

4

4

-22 0 22 Born before December 1955 (in months)

4

	He	ousehold La	bor			
	Force Survey (2019)			Analysis Sample		
Variable	Mean	S.D.	Obs	Mean	S.D.	Obs
Age	64.07	3.43	53,584	64.21	3.34	1,909
Female (%)	0.52	0.50	53,584	0.43	0.50	1,909
Marital Status (%)						
Never Married	0.02	0.12	53,584	0.02	0.13	1,907
Married	0.83	0.37	53,584	0.81	0.40	1,907
Divorced	0.03	0.17	53,584	0.03	0.16	1,907
Widowed	0.12	0.33	53,584	0.15	0.36	1,907
Education (%)						
Illiterate	0.19	0.39	53,584	0.13	0.34	1,896
Literate but no formal schooling	0.08	0.28	53,584	0.08	0.27	1,896
Primary school	0.49	0.50	53,584	0.37	0.48	1,896
Secondary school	0.06	0.24	53,584	0.15	0.35	1,896
Highschool	0.09	0.29	53,584	0.16	0.37	1,896
College and above	0.08	0.27	53,584	0.11	0.31	1,896

TABLE A1: COMPARISON OF BASIC DEMOGRAPHIC INFORMATION WITH HOUSEHOLD LABOR FORCE SURVEY

Notes: The sample includes all individuals born between January 1950 and December 1961. Age is calculated as in 2020.

	±17	±24	±30	±36	±45	±48	±60	±72
Completed bish select								
Completea nign school	0.051	0.005	0.057	0.027	0.055	0.020	0.004	0.010
Born before 1955	0.051	0.025	-0.056	-0.037	-0.055	-0.039	-0.024	-0.013
	(0.080)	(0.073)	(0.074)	(0.071)	(0.063)	(0.060)	(0.054)	(0.049)
	[0.527]	[0.733]	[0.453]	[0.602]	[0.381]	[0.520]	[0.657]	[0.791]
	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	523	692	854	1000	1246	1307	1638	1896
Control group mean	0.32	0.33	0.31	0.31	0.31	0.31	0.31	0.31
Illiterate								
Born before 1955	-0.018	-0.023	-0.004	-0.011	-0.007	-0.015	-0.030	-0.032
	(0.052)	(0.047)	(0.039)	(0.040)	(0.034)	(0.032)	(0.029)	(0.027)
	[0.738]	[0.629]	[0.928]	[0.775]	[0.835]	[0.652]	[0.297]	[0.237]
	$\langle 1.000 \rangle$							
Observations	523	692	854	1000	1246	1307	1638	1896
Control group mean	0.12	0.13	0.12	0.12	0.13	0.13	0.12	0.11
Female								
Born before 1955	0.019	0.040	0.056	0.076	0.040	0.029	0.032	0.027
	(0.080)	(0.062)	(0.065)	(0.060)	(0.056)	(0.055)	(0.050)	(0.046)
	Ì0.8091	0.518	0.3891	Ì0.2081	0.4771	0.5921	[0.521]	0.553
	(1,000)	(1,000)	(1.000)	(1.000)	(1 000)	(1.000)	(1.000)	(1.000)
Observations	525	696	859	1007	1254	1316	1650	1909
Control group mean	0.39	0 39	0 40	0.42	0.44	0 44	0.45	0.46
Married	0.09	0.39	0.40	0.42	0.11	0.11	0.40	0.10
Born boforo 1955	0.018	0.019	0.024	0.015	0.017	0.012	-0.014	-0.019
boint before 1955	(0.060)	(0.054)	(0.024	(0.045)	(0.042)	(0.012	(0.026)	(0.024)
	[0.009]	(0.034)	(0.031)	(0.043)	(0.042)	(0.041)	(0.030)	[0.034]
	[0.801]	[0.722]	[0.640]	[0.737]	[0.679]	[0.768]	[0.697]	[0.590]
	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	525	696	859	1006	1253	1315	1648	1907
Control group mean	0.83	0.84	0.85	0.85	0.85	0.85	0.84	0.84
Widowed or separated								
Born before 1955	-0.013	0.004	-0.017	-0.015	-0.021	-0.020	0.004	0.009
	(0.068)	(0.048)	(0.049)	(0.043)	(0.041)	(0.040)	(0.035)	(0.033)
	[0.851]	[0.930]	[0.721]	[0.727]	[0.603]	[0.621]	[0.918]	[0.787]
	$\langle 1.000 \rangle$							
Observations	525	696	859	1006	1253	1315	1648	1907
Control group mean	0.15	0.15	0.13	0.13	0.13	0.13	0.14	0.14
Non-Turkish								
Born before 1955	0.200	0.127	0.088	0.059	0.050	0.062	0.013	0.012
	(0.104)	(0.087)	(0.072)	(0.065)	(0.058)	(0.055)	(0.048)	(0.042)
	[0.062]	[0 153]	[0 224]	[0 369]	[0 387]	[0.260]	[0 786]	[0 780]
	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	513	682	8/3	001	1232	1294	1624	1881
Control group moon	0.22	0.26	0.25	0.26	0.26	0.26	0.25	0.25
Due Carried 10 household size	0.25	0.26	0.25	0.26	0.26	0.26	0.23	0.25
Pre-Coolu-19 nousenola size	0.1/0	0.011	0.07/	0.00	0.17(0.1.47	0.155	0.100
born before 1955	-0.162	-0.211	-0.276	-0.226	-0.176	-0.147	-0.155	-0.193
	(0.223)	(0.192)	(0.166)	(0.165)	(0.155)	(0.151)	(0.134)	(0.126)
	[0.473]	[0.278]	[0.102]	[0.175]	[0.258]	[0.333]	[0.252]	[0.127]
	(1.000)	$\langle 1.000 \rangle$	(1.000)	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	(1.000)
Observations	525	696	859	1007	1254	1316	1650	1909
Control group mean	3.38	3.39	3.34	3.37	3.39	3.40	3.41	3.40
Ever received psychological su	ıpport							
Born before 1955	-0.032	-0.069	-0.049	-0.032	-0.019	-0.018	-0.010	-0.028
	(0.051)	(0.046)	(0.042)	(0.037)	(0.034)	(0.033)	(0.032)	(0.028)
	[0.544]	[0.135]	[0.250]	[0.395]	[0.574]	[0.591]	[0.761]	[0.325]
	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	520	688	850	998	1243	1304	1634	1887
Control group mean	0.12	0.11	0.12	0.12	0.12	0.12	0.12	0.12
Has a chronic disease								
Born before 1955	-0.067	-0.026	-0.011	0.026	0.064	0.055	0.052	0.042
2011 Delote 1700	(0.071)	(0.053)	(0.047)	(0.044)	(0.043)	(0.042)	(0.036)	(0.033)
	[0.071]	[0.033]	[0.047]	[0.544]	[0.043]	(0.042) [0.197]	[0.030]	[0.000]
	(0.334)	[0.050] /1.000\	[0.024] /1.000\	[0.337]	(1.000)	[0.10/] /1.000\	[0.147]	[0.209] /1.000\
Ola a surra ti	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	522	691	853	1001	1247	1309	1640	1898
Control group mean	0.52	0.51	0.51	0.51	0.51	0.51	0.50	0.49
Joint p-value	0.26	0.25	0.39	0.59	0.53	0.51	0.75	0.58

TABLE A2: RD TREATMENT EFFECTS ON PREDETERMINED COVARIATES

Notes: This table presents RD estimates of being born before December 1955 on the predetermined characteristics of individuals. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fifth columns expand the sample to include individuals within 24, 30, 36, 45, 48, 60, and 72 months of the age threshold. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±24	±36	±48	±60	±72
Days outside last week					
Born before 1955	-1.015	-1.116	-1.037	-1.023	-0.994
	(0.236)	(0.213)	(0.170)	(0.166)	(0.153)
	[<0.001]	[<0.001]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.001 \rangle$				
Observations	672	976	1274	1601	1856
Control group mean	2.40	2.33	2.33	2.40	2.41
Under curfew					
Born before 1955	0.662	0.685	0.718	0.723	0.730
	(0.063)	(0.052)	(0.043)	(0.036)	(0.032)
	[<0.001]	[<0.001]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.001 \rangle$				
Observations	678	982	1283	1610	1866
Control group mean	0.10	0.10	0.09	0.08	0.08
Never goes out					
Born before 1955	0.304	0.317	0.285	0.245	0.281
	(0.051)	(0.038)	(0.034)	(0.034)	(0.032)
	[<0.001]	[<0.001]	[<0.001]	[<0.001]	[<0.001]
	$\langle 0.001 \rangle$	(0.001)	(0.001)	$\langle 0.001 \rangle$	$\langle 0.001 \rangle$
Observations	667	966	1264	1591	1844
Control group mean	0.16	0.19	0.19	0.18	0.18

TABLE A3: EFFECTS OF CURFEW ON MOBILITY OUTCOMES USING ALTERNATIVE BANDWIDTHS

Notes: This table presents the effects of being born before December 1955 on the mobility outcomes of individuals across alternative bandwidth selections. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 24 months of the age threshold, December 1955. The second through fifth columns expand the sample to include individuals within 36, 48, 60, and 72 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±24	±36	±48	±60	±72
Mental distress index					
Born before 1955	0.337	0.224	0.215	0.238	0.155
	(0.133)	(0.113)	(0.086)	(0.077)	(0.077)
	[0.014]	[0.051]	[0.014]	[0.003]	[0.046]
	$\langle 0.062 \rangle$	$\langle 0.118 \rangle$	$\langle 0.059 \rangle$	$\langle 0.011 \rangle$	$\langle 0.093 \rangle$
Observations	629	912	1187	1485	1725
Somatic symptoms of distress in	ndex				
Born before 1955	0.292	0.193	0.178	0.198	0.164
	(0.147)	(0.104)	(0.081)	(0.080)	(0.070)
	[0.053]	[0.066]	[0.031]	[0.015]	[0.021]
	$\langle 0.081 \rangle$	$\langle 0.118 \rangle$	$\langle 0.059 \rangle$	$\langle 0.016 \rangle$	(0.093)
Observations	666	967	1262	1580	1833
Nonsomatic symptoms of distre	ess index				
Born before 1955	0.230	0.157	0.165	0.188	0.109
	(0.126)	(0.111)	(0.085)	(0.073)	(0.071)
	[0.075]	[0.161]	[0.054]	[0.011]	[0.129]
	$\langle 0.081 \rangle$	$\langle 0.118 \rangle$	$\langle 0.059 \rangle$	$\langle 0.016 \rangle$	$\langle 0.101 \rangle$
Observations	632	916	1191	1491	1731
Sum of "yes" answers in SRQ-	20				
Born before 1955	1.045	0.876	0.751	0.816	0.487
	(0.633)	(0.491)	(0.425)	(0.384)	(0.361)
	[0.105]	[0.079]	[0.080]	[0.036]	[0.179]
	$\langle 0.086 \rangle$	$\langle 0.118 \rangle$	$\langle 0.064 \rangle$	$\langle 0.021 \rangle$	$\langle 0.101 \rangle$
Observations	629	912	1187	1485	1725
Control group mean	7.00	7.05	7.05	7.16	7.07

TABLE A4: EFFECTS OF CURFEW ON MENTAL HEALTH OUTCOMES USING ALTERNATIVE BANDWIDTHS

Notes: This table presents the reduced-form effects of being born before December 1955 on the mental health outcomes across alternative bandwidths. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 24 months of the age threshold, December 1955. The second through fifth columns expand the sample to include individuals within 36, 48, 60, and 72 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±17	±30	±45	±60
Panel A: Self-reported Healt	h Outcomes			
Poor physical health				
Born before 1955	0.140	0.119	0.103	0.070
	(0.065)	(0.046)	(0.033)	(0.034)
	[0.038]	[0.012]	[0.003]	[0.041]
	$\langle 0.083 \rangle$	$\langle 0.026 \rangle$	$\langle 0.006 \rangle$	$\langle 0.089 \rangle$
Observations	511	838	1224	1612
Control group mean	0.08	0.10	0.12	0.10
Poor mental health				
Born before 1955	0.037	0.052	0.022	-0.012
	(0.069)	(0.039)	(0.035)	(0.031)
	[0.602]	[0.192]	[0.536]	[0.701]
	$\langle 0.431 \rangle$	$\langle 0.107 \rangle$	$\langle 0.366 \rangle$	$\langle 0.540 \rangle$
Observations	509	836	1221	1607
Control group mean	0.11	0.13	0.14	0.13
Panel B: Life Satisfaction Ou	itcomes			
Suffering				
Born before 1955	-0.101	-0.005	0.000	-0.013
	(0.075)	(0.045)	(0.039)	(0.033)
	[0.184]	[0.906]	[0.990]	[0.692]
	$\langle 0.226 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$
Observations	486	801	1167	1538
Control group mean	0.21	0.22	0.23	0.23

TABLE A5: EFFECTS OF CURFEW ON SELF-REPORTED HEALTH AND LIFE SATISFACTION

Notes: This table presents regression discontinuity estimates of the effect of the curfew on self-reported health and life satisfaction measures. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	± 24	±36	± 48	± 60	±72
Panel A: Employment and	Income Outcon	nes			
Paid employed					
Born before 1955	-0.089	-0.062	-0.069	-0.064	-0.060
	(0.058)	(0.041)	(0.035)	(0.032)	(0.030)
	[0.133]	[0.140]	[0.053]	[0.044]	[0.051]
	(0.361)	(0.566)	(0.360)	(0.285)	(0.268)
Observations	673	977	1275	1597	1851
Control group mean	0.16	0.15	0.16	0.18	0.19
Paid or unpaid employed					
Born before 1955	-0.035	-0.016	-0.032	-0.039	-0.044
	(0.064)	(0.045)	(0.039)	(0.034)	(0.033)
	[0.586]	[0.722]	[0.413]	[0.263]	[0.187]
	(1.000)	$\langle 1.000 \rangle$	$\langle 0.671 \rangle$	(0.358)	(0.268)
Observations	673	977	1276	1601	1856
Control group mean	0.18	0.18	0.2	0.21	0.22
Has a job but could not attend	last week				
Born before 1955	0.005	-0.005	-0.042	-0.042	-0.045
	(0.058)	(0.042)	(0.036)	(0.032)	(0.031)
	[0.939]	[0.899]	[0.241]	[0.187]	[0.147]
	(1.000)	(1.000)	(0.474)	(0.333)	(0.268)
Observations	666	968	1268	1586	1836
Control group mean	0.16	0.16	0.17	0.18	0.18
Has enough money for usual n	eeds				
Born before 1955	-0.027	0.088	0.069	0.074	0.074
	(0.067)	(0.060)	(0.056)	(0.048)	(0.043)
	[0.686]	[0.145]	[0.216]	[0.121]	[0.084]
	$\langle 1.000 \rangle$	$\langle 0.566 \rangle$	$\langle 0.474 \rangle$	(0.320)	$\langle 0.268 \rangle$
Observations	673	978	1279	1601	1856
Control group mean	0.56	0.57	0.58	0.57	0.58
Worried about spending money	/				
Born before 1955	-0.126	-0.032	-0.021	-0.030	-0.048
	(0.062)	(0.045)	(0.044)	(0.042)	(0.040)
	[0.048]	[0.481]	[0.630]	[0.478]	[0.228]
	(0.313)	$\langle 0.927 \rangle$	$\langle 0.671 \rangle$	$\langle 0.434 \rangle$	(0.268)
Observations	672	974	1274	1597	1852
Control group mean	0.61	0.59	0.61	0.62	0.62

TABLE A6: EFFECTS OF CURFEW ON POTENTIAL CHANNELS USING ALTERNATIVE BANDWIDTHS

	±24	±36	±48	±60	±72
Panel B: Social and Physica	l Isolation Outc	omes			
Limited social interaction					
Born before 1955	0.109	0.106	0.102	0.083	0.088
	(0.065)	(0.056)	(0.050)	(0.042)	(0.042)
	[0.101]	[0.063]	[0.046]	[0.049]	[0.039]
	$\langle 0.054 \rangle$	(0.033)	$\langle 0.024 \rangle$	$\langle 0.026 \rangle$	$\langle 0.021 \rangle$
Observations	676	981	1281	1608	1864
Control group mean	0.57	0.57	0.55	0.53	0.53
Limited physical activity					
Born before 1955	0.319	0.255	0.257	0.219	0.216
	(0.071)	(0.061)	(0.052)	(0.047)	(0.045)
	[<0.001]	[<0.001]	[<0.001]	[<0.001]	[<0.001]
	(0.001)	(0.001)	(0.001)	(0.001)	(0.001)
Observations	664	965	1262	1585	1837
Control group mean	0.44	0.45	0.46	0.45	0.45
Panel C: Household Conflic	t Outcomes				
Household size					
Born before 1955	0.019	0.002	-0.018	0.013	0.000
	(0.210)	(0.183)	(0.160)	(0.146)	(0.132)
	[0.927]	[0.991]	[0.910]	[0.930]	[1.000]
	$\langle 1.000 \rangle$	$\langle 0.982 \rangle$	$\langle 0.835 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$
Observations	678	984	1285	1612	1868
Control group mean	3.55	3.50	3.52	3.54	3.52
Conflict with a household mem	ber				
Born before 1955	0.041	0.103	0.060	0.023	0.010
	(0.063)	(0.049)	(0.039)	(0.037)	(0.035)
	[0.518]	[0.038]	[0.133]	[0.523]	[0.770]
	(1.000)	$\langle 0.084 \rangle$	(0.362)	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$
Observations	662	962	1257	1579	1829
Control group mean	0.35	0.38	0.38	0.38	0.37

TABLE A6: EFFECTS OF CURFEW ON POTENTIAL CHANNELS USING ALTERNATIVE BANDWIDTHS, CONT.'D

Notes: This table presents the reduced-form effects of being born before December 1955 on the potential channels across different bandwidths. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 24 months of the age threshold, December 1955. The second through fifth columns expand the sample to include individuals within 36, 48, 60, and 72 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	±17	±30	±45	±60
Considers himself/herself religious				
Born before 1955	-0.119	-0.031	-0.022	0.028
	(0.085)	(0.047)	(0.039)	(0.033)
	[0.174]	[0.519]	[0.575]	[0.390]
	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$
Observations	494	815	1191	1573
Control group mean	0.82	0.79	0.79	0.81
Prays daily				
Born before 1955	-0.057	0.019	0.028	0.050
	(0.084)	(0.046)	(0.041)	(0.035)
	[0.500]	[0.683]	[0.493]	[0.150]
	(1.000)	(1.000)	(1.000)	(1.000)
Observations	502	819	1197	1574
Control group mean	0.67	0.67	0.67	0.68
Agrees that one should live by the holy book				
Born before 1955	-0.038	0.009	0.016	0.015
	(0.092)	(0.047)	(0.041)	(0.034)
	[0.681]	[0.852]	[0.703]	[0.66]
	(1.000)	(1.000)	(1.000)	$\langle 1.000 \rangle$
Observations	479	785	1156	1519
Control group mean	0.69	0.69	0.68	0.69
Agrees that virus is a God-sent warning				
Born before 1955	-0.093	-0.042	-0.005	0.029
	(0.100)	(0.058)	(0.047)	(0.042)
	[0.357]	[0.473]	[0.919]	[0.492]
	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$
Observations	483	790	1159	1521
Control group mean	0.58	0.56	0.56	0.58
Religiosity index				
Born before 1955	-0.180	-0.017	0.009	0.082
	(0.228)	(0.104)	(0.092)	(0.078)
	[0.435]	[0.868]	[0.919]	[0.299]
	$\langle 1.000 \rangle$	$\langle 1.000 \rangle$	$\overline{\langle 1.000 \rangle}$	$\overline{\langle 1.000 \rangle}$
Observations	461	755	1110	1458

TABLE A7: EFFECTS OF CURFEW ON RELIGIOSITY OUTCOMES

Notes: This table presents regression discontinuity estimates of the effect of the curfew on religiosity outcomes using a linear control function. The variable descriptions are provided in Appendix B. The first column presents results for individuals born within 17 months of the age threshold, December 1955. The second through fourth columns expand the sample to include individuals within 30, 45, and 60 months of the age threshold. The specification includes month fixed effects, province fixed effects, surveyor fixed effects, as well as indicator variables for education levels, ethnicity, and gender. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.

	Days outside	Under	Never		
	last week	curfew	goes out		
Born before 1955	-0.964	0.601	0.256		
	(0.246)	(0.064)	(0.053)		
	[<0.001]	[<0.001]	[<0.001]		
	(0.001)	(0.001)	(0.001)		
Observations	1214	1222	1203		
Control group mean	2.30	0.09	0.20		
	Overall	Somatic	Nonsomatic	Sum of "ves"	
	depression index	depression index	depression index	answers in SRO-20	
			acpression maex		
Born before 1955	0.278	0.253	0.173	1.004	
	(0.132)	(0.125)	(0.132)	(0.564)	
	[0.038]	[0.046]	[0.194]	[0.078]	
	$\langle 0.102 \rangle$	$\langle 0.102 \rangle$	$\langle 0.108 \rangle$	$\langle 0.102 \rangle$	
Observations	1133	1203	1137	1133	
		Paid or	Has a job but	Has enough	Worried about
	Paid	unpaid	could not attend	money for	spending
	employed	employed	last week	usual needs	money
Born before 1955	-0.042	0.012	0.023	-0.011	-0.033
boin before 1985	(0.054)	(0.058)	(0.028)	(0.086)	(0.059)
	(0.054)	(0.000)	(0.040)	[0.896]	[0.580]
	[0.441]	(0.041)	(1.000)	(1,000)	[0.380]
Observentions	(1.000)	(1.000)	(1.000)	(1.000)	(1.000)
Observations	1215	1216	1207	1218	1213
Control group mean	0.16	0.19	0.16	0.58	0.60
	Limited	Limited			
	physical	social			
	activity	interaction			
Born before 1955	0.162	0.056			
	(0.078)	(0.068)			
	[0.041]	[0.416]			
	(0.089)	(0.263)			
Observations	1201	1220			
Control group mean	0.46	0.56			
		Conflict with			
	Household	a household			
	rizo	mombor			
	5120	member			
Born before 1955	0.148	0.138			
	(0.241)	(0.066)			
	[0.542]	[0.040]			
	(0.372)	$\langle 0.088 \rangle$			
Observations	1224	1200			
Control group mean	3.50	0.38			

TABLE A8: EFFECTS OF CURFEW ON MAIN OUTCOMES USING A QUADRATIC CONTROL FUNCTION

Notes: This table presents the reduced-form regression discontinuity estimates of the effect of the curfew main outcome variables using a quadratic control function. The variable descriptions are provided in Appendix B. All columns report the reduced-form RD treatment effects of being born before December 1955 with a quadratic control function in the month-year of birth on each side of the discontinuity. The sample consists of individuals born within 45 months of the age threshold, December 1955. Standard errors, clustered at the month-year cohort level, are in parenthesis. Corresponding *p*-values and Anderson (2008)'s sharpened *q*-values are in square and angle brackets, respectively.